

Review



Wearable Sensors and Smart Devices to Monitor Rehabilitation Parameters and Sports Performance: An Overview

Roberto De Fazio ^{1,2,*,†}, Vincenzo Mariano Mastronardi ^{1,3,*,†}, Massimo De Vittorio ^{1,3} and Paolo Visconti ^{1,3}

- ¹ Department of Innovation Engineering, University of Salento, 73100 Lecce, Italy
- ² Facultad de Ingeniería, Universidad Panamericana, Aguascalientes 20290, Mexico
- ³ Center for Biomolecular Nanotechnologies, Italian Technology Institute IIT, 73010 Arnesano, Italy
- * Correspondence: roberto.defazio@unisalento.it (R.D.F.); vincenzo.mastronardi@unisalento.it (V.M.M.);
 - Tel.: +39-08-3229-7334 (R.D.F.) Both the authors are to be considered as first author.

Abstract: A quantitative evaluation of kinetic parameters, the joint's range of motion, heart rate, and breathing rate, can be employed in sports performance tracking and rehabilitation monitoring following injuries or surgical operations. However, many of the current detection systems are expensive and designed for clinical use, requiring the presence of a physician and medical staff to assist users in the device's positioning and measurements. The goal of wearable sensors is to overcome the limitations of current devices, enabling the acquisition of a user's vital signs directly from the body in an accurate and non-invasive way. In sports activities, wearable sensors allow athletes to monitor performance and body movements objectively, going beyond the coach's subjective evaluation limits. The main goal of this review paper is to provide a comprehensive overview of wearable technologies and sensing systems to detect and monitor the physiological parameters of patients during post-operative rehabilitation and athletes' training, and to present evidence that supports the efficacy of this technology for healthcare applications. First, a classification of the human physiological parameters acquired from the human body by sensors attached to sensitive skin locations or worn as a part of garments is introduced, carrying important feedback on the user's health status. Then, a detailed description of the electromechanical transduction mechanisms allows a comparison of the technologies used in wearable applications to monitor sports and rehabilitation activities. This paves the way for an analysis of wearable technologies, providing a comprehensive comparison of the current state of the art of available sensors and systems. Comparative and statistical analyses are provided to point out useful insights for defining the best technologies and solutions for monitoring body movements. Lastly, the presented review is compared with similar ones reported in the literature to highlight its strengths and novelties.

Keywords: advanced diagnostics; bio-vital markers; e-healthcare; injury-prevention; mini-invasive monitoring; rehabilitation parameters; sports performance tracking; wearable sensors

1. Introduction

Monitoring human physiological functions and performance during real-time activities is becoming increasingly popular in sporting and healthcare environments [1–4]. Recent studies demonstrated that introducing enhanced recovery after surgery programs can significantly improve post-surgical recovery quality, thereby helping to decrease the time for rehabilitation outside clinical centers [5,6]. Traditionally, doctors or physicians assist the patients in completing the rehabilitation training using hands or basic and standard equipment. The assessment of a patient's physical state, the degree of rehabilitative training, and the effectiveness of the rehabilitation results are thus highly dependent on the physician's expertise and level. Moreover, more than 70% of patients usually fail to comply with post-surgical programs when moving home. The main limitations to a successful



Citation: De Fazio, R.; Mastronardi, V.M.; De Vittorio, M.; Visconti, P. Wearable Sensors and Smart Devices to Monitor Rehabilitation Parameters and Sports Performance: An Overview. *Sensors* 2023, 23, 1856. https://doi.org/10.3390/s23041856

Academic Editor: Michael Beigl

Received: 12 January 2023 Revised: 2 February 2023 Accepted: 3 February 2023 Published: 7 February 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). rehabilitation are the lack of time and feedback from specialists, the difficulties in remembering how to execute the exercises, and the inability to feel the benefits of the therapies and the related improvements. Therefore, the measurement of a human being's kinetic and other vital and physiological parameters is essential to monitor and collect a series of data (e.g., movements, bending, joint rotation, and so on) useful to provide important tracking feedback to the user. This allows for the adoption of suitable protocols that ensure the patients carry out their physiotherapy programs even at home. In rehabilitation after trauma or injuries, these data allow for defining the correct therapies and evaluating their effectiveness and patient progress. Moreover, in sports training, they allow for optimizing the exercises and assessing an athlete's progress.

The rapid development of the Internet of things [7–9], due to the technological advancement of sensor and communication technology, together with increased availability, lower cost, and advancements in personal computing devices, has paved the way for new opportunities and challenges for the research and development of wearable healthcare devices [7,8,10–14]. Wearable sensors enable users to identify and measure different clinically relevant parameters, such as functional motions, biomechanical and bio–vital parameters, and athletes' workloads, to enhance performance while reducing the risk of injury. Wearable monitoring devices can give a continuous and real-time flow of physiological data, creating precise treatment plans and player-specific training regimens to help minimize or prevent injuries.

This review aims to provide an overview of the most recent wearable sensors and systems in the literature used to detect and monitor physiological parameters in athlete training tracking and post-operative rehabilitation of patients. At first, the most significant parameters for monitoring post-operative rehabilitation and sports performances are introduced. Additionally, a first classification of the discussed and analyzed parameters is provided, distinguishing between medical rehabilitation and sports fields. After this, an analysis of the available technologies for detecting and acquiring the physiological parameters directly from the human body is provided. In detail, wearable sensors will be classified according to the parameters they allow to identify, such as body movements or physiological parameters. Inertial, capacitive, piezoresistive, piezoelectric, and flexoelectric transduction mechanisms will be detailed. Afterward, some figures of merit (FoM) helpful for classifying the analyzed technologies in terms of performance are shortly described. FoM are used in this paper to identify the characteristic parameters of the discussed technologies and compare and classify the working mechanisms used to implement the analyzed transduction mechanisms. These quantities allow quantifying their efficiency and performance. An overview of wearable technologies for monitoring body motions is presented, providing a detailed depiction of the available sensors and systems. Additionally, they are classified in two main categories: (i) wearable sensors for post-operative rehabilitation monitoring and (ii) wearable sensors for athletes' performance monitoring. The described technologies are then compared to achieve a detailed classification according to the sensor's maturity, sensitivity, the range of detection, the area of the body where they are typically placed, the measured parameters allowed by that specific technology, and the corresponding response time. In addition, several publicly available datasets related to sports and rehabilitation monitoring are discussed. Finally, a comparison of the presented review work with similar ones reported in the scientific literature is presented to highlight its novelties and strengths. The main novelties and contributions of the presented review are:

- A detailed classification of human physiological parameters useful for extracting information about a patient's health status during post-operative rehabilitation and sports performances.
- An in-depth analysis of the transduction mechanisms for acquiring parameters related to body motions; also, innovative wearable technologies and sensors to monitor human activities are discussed, ranked by high comfort and flexibility.

- A comprehensive overview of available wearable technologies for monitoring motions of body parts and physiological parameters to enhance rehabilitation therapies and customize athletes' training.
- A comparison of the presented review with similar ones reported in the literature; its strengths lie in its completeness and level of detail, dealing with both sensing systems for monitoring rehabilitation and sports performances. In addition, it does not limit the discussion to specific sensor categories, applications, or monitored body areas, as detailed in the comparison reported in Section 6. The joint discussion of the two applications represents one of the novelties of the presented work, rarely treated in other review articles with the presented level of detail. In addition, the proposed work reports comparative analyses related to the discussed scientific studies from the performance point of view, providing useful insights for determining the best sensing strategies for developing future wearable systems for monitoring the human body.

The remainder of the review is arranged as follows: Section 2 presents a detailed classification of the primary parameters acquired on the body surface to extract information related to post-operative rehabilitation or sports training. Afterward, Section 3 analyzes the main sensing mechanisms for detecting body motions related to physiological activities; later, the figures of merit for comparing and classifying sensing mechanisms implemented by wearable devices are described. Section 5 reports a survey of available wearable technologies for monitoring body motions which find application in post-operative rehabilitation and athlete performance tracking. Furthermore, an overview of datasets related to sports and rehabilitation monitoring is introduced. Lastly, Section 6 presents a comparative analysis of the presented review work with similar ones reported in the literature, highlighting its strengths and novelties.

Selection and Exclusion Criteria for the Presented Review Paper

Before starting the discussion of the presented review is necessary to define the criteria used to select and exclude the most appropriate scientific works and review papers. The latter has been defined specifically, taking into account many elements of the analyzed documents, such as applicability to the treated themes, relevance, publication year, and redundancy concerning other chosen articles. The goal was to provide the reader with as broad a view as possible of wearable systems and technologies for monitoring body motions and biophysical parameters which assist medical staff in a patient's rehabilitation process and tracking an athlete's performances. The selection process was conducted following the methodology shown in Figure 1; in particular, analysis of the documents was performed according to a three-step procedure. In detail, the procedure began with the title's evaluation, followed by the analysis of the abstract, then coming to the careful reading of the full paper. Each step involved a binary evaluation, leading to acceptance or rejection of the considered document according to the criteria above (Figure 1). At the end of the analysis, if the document was unclear, analysis of its contents was further deepened by researching information from external sources. If, after this study, the contents remained unclear, the document was discarded from the review, noting the reasons for its exclusion.

Ultimately, 141 documents were analyzed to realize the overview of smart devices for monitoring rehabilitation and tracking sports activities; Figure 2 depicts the distribution of the selected documents grouped according to their typology (research articles, review articles, and books).

Furthermore, Figure 3 reports the distribution of the databases and the main keywords used to research the documents included in the presented review.



Figure 1. Workflow describing the selection process involved in the presented review paper.



Figure 2. Distribution of the documents selected for structuring the presented review paper on smart devices for monitoring rehabilitation and tracking sports activities.



Figure 3. Distribution of the databases (**a**) and main keywords (**b**) used for searching the documents included in this review.

2. Classification of Human Physiological Parameters

The measurement of vital human parameters is essential to provide important tracking insights and feedback on the health status of the final user. For this reason, the acquisition of information directly from the human body by sensors attached to sensitive skin locations or worn as a part of garments allow obtaining clinically relevant parameters. They ensure a continuous flow of data, going beyond the limits of traditional rehabilitation and training methods [15].

Some joints, such as the elbow, are limited to bending in one direction; others, such as the shoulders and hip, can perform more complex movements in three dimensions. In this respect, one of the relevant key parameters that should be monitored during rehabilitation after injury to the bones, muscles, or nerves is the range of motion (*ROM*) [16–18]. Being calculated as the joint's rotation angle in respect to a reference plane, as in the case for elbow flexion/extension, or a reference axis, as in forearm pronation/supination movements, it is defined as the freedom of joint movements in a space (Table 1). It is obtained by measuring the angle between the standard joint position and the final one the patient can reach without experiencing pain; thus, it is a suitable parameter to evaluate improvements over time [19–23].

 Table 1. Table showing the main angles on the joints of the upper and lower body limbs.

Symbol	Representation
$\theta_{S-ab/ad}$	Shoulder abduction/adduction
$\theta_{\rm S-hab/ahd}$	Shoulder horizontal abduction/adduction
$\theta_{S-in/ex}$	Shoulder internal/external rotation
$\theta_{\rm W-fl/ex}$	Wrist flexion/extension
$\theta_{W-ra/ul}$	Wrist radial deviation/ulnar deviation
$\theta_{\rm F-pr/su}$	Forearm pronation/supination
$\theta_{\rm E-fl/ex}$	Elbow flexion/extension
$\theta_{A-IC, \text{ front}}$	Ankle initial contact angle—frontal plane
$\theta_{A-IC, sag}$	Ankle initial contact angle—sagittal plane
$\theta_{A-MAX, front}$	Ankle maximum—frontal plane
$\theta_{A-MAX, sag}$	Ankle maximum angle—sagittal plane
$\theta_{K-IC, \text{ front}}$	Knee initial contact angle—frontal plane
$\theta_{\rm K-IC, sag}$	Knee initial contact angle—sagittal plane
$\theta_{K-MAX, front}$	Knee maximum—frontal plane
$\theta_{K-MAX, sag}$	Knee maximum angle—sagittal plane

In this context, Kim et al. in [24] analyzed patients' upper joint movement ranges by using wearable devices. As a monitoring benchmark, the authors calculated the angles of different upper joints. The wrist flexion/extension, wrist radial/ulnar deviation, shoulder external/internal rotation, shoulder adduction/abduction, forearm supination/pronation, and elbow flexion/extension were then studied by tracking the detected angles, enabling to emulate each patient's movements using a real-time animation of an avatar to improve the monitoring experience (Figure 4a). An additional parameter that allows evaluating a patient's improvements during rehabilitation is the comparison between the mobility range of the limb affected by the trauma and the mobility of a healthy one—e.g., how much the injured limb is used during the day—establishing the movement frequency. Together with the parameters above, the patient's balance and gait [25–27], cardiac and respiratory parameters (such as HR—heart rate, blood pressure, HRV—heart rate variability, RR—respiration rate, etc.) [28-31], neck muscle vibrations (frequency, pattern, intensity) [32], chewing (duration and frequency) [21], and swallowing frequency [33,34] are recorded. The latest, especially in patients affected by diseases such as dysphasia and dysphagia, are studied and analyzed to monitor the patient's health after surgery and understand how some pathologies affect their normal state.



Figure 4. Schematic illustration of a subject wearing tendon-inspired motion detection devices and pictures of the integrated device into wearable belts (**a**) [24]. The main angles measured on the lower limbs for rehabilitation monitoring and sports tracking are highlighted in (**b**).

As reported in Table 1, other fundamental angles can be extracted for the lower limb. The knee and ankle angles are essential for preventing injuries; in particular, the initial contact angle (IC), maximum angle (MAX) at the midstance, as well as the latency between the IC and MAX can be extracted both in the frontal and sagittal planes (Figure 4b) [35]. These parameters are subject to significant variations during running over long distances, and are also affected by the typology of worn footwear [35]. Furthermore, from the analysis of kinetic data, the elastic characteristics of ligament structures in the knee extensors and plantar flexors can be calculated [36].

As with the post-operative rehabilitation process, it is also possible to identify some key physiological parameters for monitoring athletes' sports performance. Some of these parameters can overlap with those mentioned above: for example, ROM and joint motion angles can provide meaningful feedback on the definition of movement strategies for athletes or the correct execution of motion in a specific discipline [37–42]. Intensity and trajectory of body movements, in addition to heart and breathing parameters (e.g., HR, SpO₂,

RR, etc.) during long physical activity periods, can carry information about an athlete's fatigue and fitness decrease [10], which provide essential feedback on the athlete's health status. Correct execution of running or jumping [43–45], hit pressure [46] in combat sports, response time, and acceleration to execute a movement [47] are additional parameters of interest that are usually monitored during sports activity tracking [48–52]. S. Saponara designed and proposed a wearable device capable of monitoring some of an athlete's key movement parameters in combat sports [47]. In this work, the author analyzed the pressure of kicks and punches on a target body to quantify a movement's effectiveness from a traumatological model correlated with the subjective indication of a coach. The acceleration of movements was also analyzed and compared with the ideal values for that specific technique, allowing the athlete to avoid trajectories that deviate from the standard. Additionally, the user's response time was studied to examine the time to pull a kick or a punch—roughly a tenth of second—depending on the combat technique and athlete's physical characteristics.

A classification of the parameters (or group of parameters) discussed above is summarized in Table 2, where medical rehabilitation and sports fields were distinguished; this classification derives from the analyses in the scientific literature reported in this review work. However, some parameters are involved both in rehabilitation monitoring and sports performance tracking, such as heart and breathing parameters. This classification aims to give the reader a comprehensive overview of the main parameters monitored during medical rehabilitation and sports activities.

Table 2. Classification of human physiological parameters usually identified in monitoring patients' rehabilitation (**a**) and athletes' activities (**b**).

Rehabilitation Parameters	Sports Activity Parameters
Joint bending (angle or ROM) ^(A) [18,21,22,24–29,32,39–44,50,53–60] ^(*)	Joint bending (angle or ROM) ^(A) [18,21,22,24–29,32,39–44,50,53–60] ^(*)
Joint rotation ^(B) [18,21,24] ^(*)	Joint rotation ^(B) [18,21,24] ^(*)
Neck muscle vibration (frequency and pattern) [22,27–30,32,39,42,53]	Jumping (distance and peak acceleration) [39,44,50,55]
Heart parameters (HR—heart rate, HRV—heart rate variability, blood pressure) [21,27,28,30,42,56] ^(*)	Heart parameters (HR—heart rate, HRV—heart rate variability, blood pressure) [21,27,28,30,42,56] ^(*)
Breathing parameters (RR—respiration rate, inhalation/exhalation times, flow rate) [21,29–32,53,61] ^(*)	Breathing parameters (RR—respiration rate, inhalation/exhalation times, flow rate) [21,29–32,53,61] ^(*)
Swallowing (interval and frequency) [21,33,34]	Hit acceleration [47]
Chewing (duration and frequency) [21]	Hit pressure [40,43,46,47]
Movement response time ^(C) [47] ^(*)	Movement response time $^{(C)}$ [47] $^{(*)}$
Movement trajectories [47] (*)	Movement trajectories [47] (*)
Walking (speed and stance time and pressure) [22,29,39,41,43,44,50,55,62] ^(*)	Walking (speed and stance time and pressure) [22,29,39,41,43,44,50,55,62] ^(*)
Plantar pressure distribution [50,62,63] (*)	Plantar pressure distribution [50,62,63] (*)
	Fatigue [64]
	Running (speed, pace, acceleration) [22,39,41,43,45,50,55,64]
(a)	(b)

^(*) It is applied for both rehabilitation and sports activities. ^(A) Bending movement around a joint (such as the knee or elbow). ^(B) Twisting movement produced by the superposition of small rotational movements around the joint. ^(C) Time interval between the stimulus' appearance and the corresponding muscle reaction's start.

In conclusion, considering the parameters and physiological processes reported in Table 2, the set measured depends on the typology of clinical rehabilitation or sports activity to be monitored. For instance, Figure 5 depicts some examples with related monitored parameters and activities.



Figure 5. Plots representing the parameters and gestures monitored for the different typologies of clinical rehabilitation: post-stroke (**a**), post-heart attack (**b**) and rehabilitation from an injury to the lower limbs (**c**). Parameters and gestures monitored in sports activity: running monitoring (**d**) and high jump tracking (**e**).

3. Transduction Mechanisms for the Acquisition of Human Parameters from the Body

In the last decade, the advancement of various sensing technologies and the development of highly flexible and conformable wearable devices and sensors have paved the way for innovative solutions that monitor human activities by exploiting minimally invasive and comfortable devices.

In broad terms, wearable sensors can be classified by considering the parameters they should identify: body movements (position, gait, acceleration, etc.) or physiological parameters (heart rate, heart rate variability, voice, etc.). However, a more detailed classification is based on the working transduction mechanism. In this respect, inertial, optical, and angular sensors are the main instruments for monitoring the human body's gait.

Optical fiber sensors (OFS) are based on optical technology and do not suffer from electromagnetic interference. OFS are composed of a light source that transmits a light beam to a photodetector through an optical fiber. They can be used to assess a joint's bending angles by measuring the angle and the attenuation of the reflected light beam when detected by the photodetector [65–68]. Joint motion angles are also measured using angular sensors or goniometers based on strain gauges or resistive potentiometers. However, some of the main problems of these sensors are their lower accuracy and rigidity, which do not allow them to be positioned comfortably on the joints. Markerless and marker-based motion capture technologies are also available for monitoring human physiological parameters.

They are used to quantify the kinematics of a motion, with the ability to enhance clinical evaluations of function and performance [25,69–74].

An alternative to the previously mentioned sensors, inertial sensors, are widely used by clinicians to perform kinematic measurements to monitor both healthy and pathological movements, quantify the degree of impairment and the severity of the damage, plan rehabilitation strategies, and evaluate the impact of various therapies. An inertial sensor consists of a small and rigid central body, the inertial measurement unit (IMU), which usually integrates MEMS devices such as accelerometers, gyroscopes, and magnetometers, enabling the perception of movement in multiple dimensions using a single sensor. They are usually placed above and below the joints (neck, fingers, elbow, shoulder, hip, knee, ankle, etc.), and they allow detection of bending angle, linear 3D acceleration, threedimensional orientation, and angular velocity to track joint movements, walking speed, etc. (Figure 6a) [75-78]. Moreover, an additional advantage of IMU is the capacity to measure a patient's or an athlete's energy consumption—critical for determining, for example, the intensity of the training—by multiple integrations of vertical acceleration over time. Despite their high accuracy, low-cost design, and portability, IMU sensors are sensitive to electromagnetic interference and noise, especially indoors; they are affected by a drift effect and high rigidity, limiting their daily application. For this reason, their utility is often restricted to clinical applications under expert supervision.



Figure 6. Schematic illustration of (**a**) inertial sensors ("reproduced with permission from S. Z. Homayounfar et al., SLAS Technology; published by Elsevier, 2020" [79]); (**b**) piezoresistive (**i**), piezoelectric (**ii**), capacitive (**iii**), and triboelectric (**iv**) sensors [80]).

Electromechanical sensors based on different transduction methods, such as capacitive, triboelectric, piezoresistive, and piezoelectric effects, are more appropriate for wearable applications. By virtue of the material used for their fabrication, these sensors are usually lighter, comfortable, robust, and suitable to detect physiological parameters directly from the human body since they can be attached to or integrated into clothes, wraps, yarns, etc. [81]. These devices can be fruitful in several healthcare applications, covering continuous health monitoring, daily and athletic activity tracking, and acting as a multifunctional electronic skin. However, the employment of nanomaterial-enabled wearable sensors on a large scale faces several challenges and technical issues, such as improving their perfor-

mance (biocompatibility, performance stability, multi-modal sensing capability), efficient integration, and advances of other compliant/stretchable components.

Piezoresistive sensors (Figure 6b-i) are the most widely used technologies to transform deformation stimuli into electrical signals. The working principle of these sensors relies on the relation

$$R = \frac{\rho l}{A} \tag{1}$$

where l is the surface's length, A the cross-section's area, and ρ the material's electrical resistivity. By choosing suitable materials, and varying one or more of these parameters by deforming the structure of the sensor, the resistance value changes accordingly. By calibrating the sensor and correlating the resistance variation to the applied compression or traction force, it is possible to evaluate the intensity and magnitude of the external stimuli. Piezoresistive sensors can be classified into strain and pressure sensors depending on the mechanical force applied. The choice of suitable materials, the production process, and the general operation of piezoresistive strain sensors have been extensively studied by researchers.

Moreover, structured 3D patterns, such as micro-pillars, half-spheres, or pyramid shapes, can affect the performance of piezoresistive sensors, as well as the combination of conductive materials such as carbon nanotubes (CNTs), cellulose nanocrystals (CNCs), silver nanowires (AgNWs), graphene, and other polymers, with a rubber body to provide elastic properties. If subjected to a tensile force, the resulting compound varies the conductive paths, redistributing the percentage of the conductivity inside the material and, consequently, varying the resistance value. An interesting fabrication method involves synthesizing the piezoresistive material as a fabric in clothes and optimizing its functionality, for example using elastic fibers painted with conductive substances or wool fibers covered with graphene nano-sheets (GNPs) or conductive carbon (CB) [27,30,32,56,82]. Yao et al. in [83] presented a novel methodology for integrating multifunctional e-textiles without sacrificing stretchability, wearability, or washability. The material was obtained by depositing from a AgNW solution onto a glass plane and evaporating the residual solvent. After, a TPU layer was spin-coated over the AgNW network, letting the solvent evaporate and solidify. Then, the layer was transferred onto a stretchable fabric by heat-pressing at 140 °C. The test results indicated that the resulting AgNW–TPU layer is characterized by low sheet resistance (0.2 Ω/\Box), a low Young's modulus, excellent stretchability (50%), and good washability (10% resistance increase after 100 washing cycles).

Capacitive sensors exploit the physical properties of a capacitor made of proper elastic materials, to vary the capacitance value when pressure or traction is applied (Figure 6b-ii). They consist of two metal electrodes separated by a dielectric elastic material, such that the capacitance is

$$C = \frac{\varepsilon_0 \varepsilon_r A}{d}$$
(2)

where A is the surface area of the two metal plates, d is the thickness of the dielectric material between them, ε_0 is the electrical permittivity of the vacuum, and ε_r is the relative electrical permittivity. The capacitance variation is then strictly related to a change in the thickness of the dielectric material or the metal plate's surface area, and it varies linearly with the applied force or pressure. Using proper readout circuitry, it is possible to transduce this capacitance variation in a voltage output signal. Despite a good sensitivity, capacitive sensors show a limited detection range, mainly due to a decrease in the elasticity of the final device when metal electrodes are deposited on the dielectric elastomer. For this reason, in a wearable application, the conductive electrodes are usually made of elastic materials such as a mixture of polydimethylsiloxane (PDMS) or EcoFlex and silver nanowires (AgNWs) [55,61], liquid indium gallium arsenide (GaInAs) alloy paints [57], or conductive hydrogels [28].

The direct piezoelectric effect (Figure 6b-iii) allows transducing the mechanical energy caused by deformation or strain into electrical energy, usually collected as a voltage

signal [84,85]. This phenomenon is based on the property of some ceramic crystals and polymeric materials to generate dynamic charge separation and a consequent electric potential due to reorientation of the dipoles inside the material lattice while subjected to a

tential due to reorientation of the dipoles inside the material lattice while subjected to a force or mechanical stress. Materials used to produce piezoelectric sensors can be organic, such as polyvinylidene fluoride (PVDF), polyacid(D-lactic) (PDLA), and polyacid(L-lactic) (PLLA), or inorganic such as aluminum nitride (AIN), lithium niobite (LiNbO₃), barium titanate (BaTiO₃), lead zirconate titanate (PZT), zinc oxide (ZnO), and quartz. Organic polymers are often preferred for their biocompatibility, simpler manufacture, and low cost. However, they show a lower piezoelectric response when compared to their inorganic counterpart [86]. One of the most used materials for piezoelectric wearable sensors is PVDF which has excellent physical properties, high polarization, and good flexibility, often combined with tetrafluoroethylene (TrFE) to improve the piezoelectric coefficients [87,88].

The triboelectric effect is a contact electrification phenomenon whereby a voltage signal is generated by transferring electric charges between two materials when they are in contact with each other and are subjected to friction and rubbing (Figure 6b-iv)—during the rubbing, one material gives electrons, charging itself positively. In contrast, the other material receives the released electrons, charging itself negatively as a consequence. Triboelectricbased sensors can be classified into four categories, according to their operating modalities: (i) vertical contact-separation sensors; (ii) in-plane sliding sensors; (iii) single-electrode sensors; (iv) freestanding triboelectric sensors [43,89–91]. Triboelectric sensors add the potential advantage of being self-powered since the output signal generated by the electrification process can be harvested and stored in batteries. For this reason, triboelectric sensors are widely employed in wearable applications, ensuring long-lasting operations such as jointmovement tracking and breathing monitoring [26]. Different materials can be used to fabricate triboelectric charge-generating and charge-trapping layers and electrodes. Usually, they are polymer-based, highly flexible materials, allowing the device to be embedded inside clothing or placed conformably in direct contact with the skin [92]. Poly(dimethylsiloxane) (PDMS), fluorinated ethylene propylene (FEP), fluorinated poly(ethylene terephthalate) (PET), and Poly(vinylidene fluoride-co-trifluoroethylene) (PVDF-TrFE) are some of the most used materials for the charge generating layer [93]. Polystyrene (PS), polyimide (PI), and PVDF, having many trapping sites due to the non-uniform energy levels along their main chains, are often used as charge-trapping layers. In some cases, they include nanoparticles, such as Ag nanowires (AgNWs), which promote trapping the electrostatically induced charges at the metal-dielectric interfaces. Solid and flexible electrodes, including metals, graphene, and indium tin oxide (ITO), are usually used for the charge-collecting layer together with conductive elastic nanocomposites, which have stable electrical properties in various mechanical environments.

A combination of the previously mentioned transduction methods is often developed to improve the response of the final hybrid sensor to the applied stimuli. Hybrid devices made by combining triboelectric and piezoresistive effects or triboelectric and piezoresistive transduction methods is also available, especially for monitoring joint movements [59], gait characteristics [63], or respiratory activity [94].

A summary of the electromechanical transduction technologies discussed above is reported in Table 3. Here a classification of the sensors and systems has been carried out, taking into account the transduction mechanism, the ability of the sensors to generate a signal without driving it by an external source (hereafter referred to as active/passive element), strengths, main drawbacks, and area of the body where the sensor can be applied.

Transduction Technology	Active/Passive	Strengths	Drawbacks	Body Area
Inertial	Passive	 Reliability High accuracy Small size Low power Wide operative range Wide bandwidth Linearity 	 Sensitivity to EMI Sensitivity to undesired body movement Complexity in data processing 	Knee [20,54,95], elbow [96,97], chest [97,98], wrist [97,99], shoulder [97], head [97], throat [100]
Piezoresistive	Passive	 Robustness Simple structure Low cost Well-known mechanism Stable over time High durability 	 High power consumption Sensitive to environmental conditions Difficulty in scaling down Poor linearity 	Foot [62], wrist [27,30,44,56,60], finger [27,32,44,56,60], throat [27,30,32], chest [31,32], elbow [32,44,60], knee [25,32,44,60], neck [32], nose [30], cheek [27,30], ankle [44], abdomen [60]
Capacitive	Passive	Low costWide operative rangeWide temperature range	 Sensitive to environmental conditions Poor linearity Vibration sensitivity 	Finger [28,29,53,55], throat [28,29,53], wrist [28,57], cheek [29], knee [29,55], elbow [29,57], foot [29], belly [53,61], head [46]
Piezoelectric	Active	 Active Robustness Simple structure Biocompatibility (according to used material) Simple scaling-down 	 Responsive only to dynamic stimuli High sensitivity to temperature and moisture 	Wrist [21,24,39,87], throat [21,22,25,28,39,87], cheek [21], nose [21,87], shoulder [24], forearm [24], elbow [22,24,39,42], foot [22], knee [39,64], finger [30,33,79], ankle [64]
Triboelectric	Active	 Active High sensitivity Adaptable to different sizes and shapes Low cost Simple structure 	 Difficult signal reading Sensitive to environmental conditions Low durability Limited short circuit output current 	Tongue [33], finger [50], wrist [26,50,91], elbow [26,41,50,58], foot [41,50,91], knee [26,41,91], belly [91], ankle [26], shoulder [26],

Simple structure

Table 3. Comparison of the main transduction technologies used in wearable systems to monitor sports and rehabilitation activities.

Establishing a comparison between the various technologies to better understand how they could work in post-operative rehabilitation and sports performance monitoring proved to be a complex task. To help the reader understand the complexity of the scenarios for these technologies, we present two summarizing figures. The radar charts of Figure 7 report a statistical analysis of the analyzed papers to determine the ability of that sensor/technology to detect frequency variations in the desired physiological activity, e.g., breathing, bending, walking, running frequency, and so on. Specifically, the "frequency" parameter was introduced for this purpose. It represents the number of analyzed papers (a total of 112 documents) where a given sensing technology was employed to monitor specific body areas (i.e., ankle, neck, chest, foot, etc.). The resulting paper numbers were normalized on a scale from 0 to 10, considering the range of the analyzed sensing technologies and the considered application areas. In addition, the analysis was carried out taking into account just the main application area on the body suggested by the authors of each paper.

output current

spine [26]



Figure 7. Radar charts for comparison of the main transduction mechanisms to monitor human bio–physical parameters according to the area of sensor application on the body. The scale (0–10) indicates the "frequency" with which a specific sensing technology is applied for monitoring a given body area. Additionally, the analyzed population (N) is reported for each sensing technology.

Additionally, the radar charts in Figure 8 depict the application rate of the technology/sensors analyzed in monitoring typical biophysical parameters, such as joint bending angles and frequency, extremity pressure, heart pulse, and throat vibration. Similarly, the "frequency parameter" was calculated to evaluate the propensity of a given sensing technology to detect the considered biophysical parameters. Similarly, in this case, only the main acquired parameter was selected for each analyzed paper.

These radar charts intend to provide a comparative graphical analysis of all the devices and sensors analyzed in this review paper. They were built considering the normalized number of cited research works whose developed technology—capacitive, triboelectric, piezoresistive, piezoelectric, and inertial—is applied to a specific body area and the desired parameter which can be detected and monitored.

As can be noted in Figure 7, piezoresistive and capacitive sensors are widely used for monitoring finger and wrist movements. Their employment can be justified by the availability of textile-based sensors, widely used for these applications since they feature a very low profile and high flexibility. However, both sensors' typologies are passive and thus require the application of polarization, which is a drawback for low-power applications. Furthermore, piezoelectric sensors are commonly applied for detecting throat movements. Indeed, several examples of thin-film piezoelectric layers are proposed in the scientific literature, showing high sensitivity, flexibility, and biocompatibility, enabling discreet and safe detection of body movements.



Figure 8. Radar chart comparing the considered transduction mechanisms according to the main detected biophysical quantities. The scale (0–10) indicates the "frequency" with which a specific sensing technology is applied for monitoring the specific parameter.

In addition, from Figure 8 it is evident that piezoresistive, capacitive, and piezoelectric sensor technologies are currently dominant for detecting joint movements (angles and frequency), as well as throat vibrations, i.e., they are applied to voice recognition. Triboelectric sensors are commonly applied to detect pressures from different body parts—e.g., feet and hands. Nevertheless, this sensor technology is not mature enough to be applied to large-scale devices. In conclusion, from the reported analysis, inertial sensors are used to a lesser extent than the other sensor technologies, probably due to their high sensitivity to extraneous body motions, inducing motion artifacts.

4. Figures of Merit for Performance Comparison

Different characteristic parameters can be identified to compare and classify the working mechanisms used for implementing the wearable sensors and systems discussed in Section 3 and quantify their efficiency and performance. These parameters will be used as figures of merit (FoM) for carrying out comparative analysis of the sensing devices reported in Section 5.

One of the most important parameters is the sensitivity (S), defined as the ratio between the variation in the output signal dX and the related variation in the input dP:

9

$$S = \frac{dX}{dP}$$
(3)

states that the input stimulus (compression or traction) generates a significant output variation in a sensor with high sensitivity [101]. However, in some cases, it is useful to

resort to the concept of normalized sensitivity to take into account in this parameter the sensor's resting value (X_0) [102]:

$$S = \frac{\Delta X / X_0}{dP}$$
(4)

A second parameter useful to describe the performance, especially of strain piezoresistive sensors, is the gauge factor G_f defined as

$$G_{f} = \frac{\Delta R/R_{0}}{\Delta L/L_{0}}$$
(5)

which is the ratio between the normalized resistance variation ΔR compared to the resting resistance R_0 as a function of the elongation variation ΔL of the resistive stripe compared to the resting length L_0 . Typical values of this parameter range between 2 and >100, depending on the shape and type of material comprising the stain sensor [103].

Extensibility E, also called Young's modulus, is an intrinsic feature of the material used. It defines the maximal threshold beyond which the ratio between the strain and the traction applied to the material causes the sensor to lose sensitivity and linear behavior. The Young's modulus is calculated according to the following equation:

$$\mathbf{E} = \frac{\mathrm{d}\sigma}{\mathrm{d}\varepsilon} \tag{6}$$

where ε is the strain generated when a traction (or stress) σ is applied. A Young's modulus approaching the elastic properties of the skin is highly desirable, especially for wearable devices in direct contact with soft tissues [104].

Additional parameters are evaluated to define the performance of wearable sensors. The range between the minimum and maximum values the wearable sensor can detect is defined as the range of detection (RoD); beyond the maximum value, the sensor is no longer sensitive to input variations [105]. The sensor response time is the time interval between the application of the external stimulus and the generation of the transduced output signal (a critical factor in dynamic and real-time activity tracking) [106].

Linearity is the percentage of deviation of the measured signal from the regression line and allows for evaluation of the signal stability in specific applications [105]. Among the other FoM, it is certainly worth mentioning the precision, which indicates the capability of the sensor to keep the measurements reproducible. Another fundamental FoM is the resolution intended as the lowest observable incremental change of an input parameter. Resolution can be represented in absolute terms or as a percentage of the reading (or the full-scale reading). Furthermore, the greatest deviation between the sensor measurement against the reference obtained by the main or a good reliable standard is defined as accuracy [105]. Furthermore, the offset error is the deviance in the sensor output from a reference value at the detection range's bottom end. Lastly, power consumption is critical in wearable applications, especially for long monitoring of daily life.

5. Overview of Available Wearable Technologies for Body Motion Monitoring

Wearable devices for monitoring body part movements and physiological parameters improve standard and traditional rehabilitation therapies since they can provide important feedback to the users—both patients and therapists—additionally allowing to customize the training path of the athletes. Section 5 reports an overview of the most recent and innovative wearable sensors and systems based on the transduction technologies detailed in Section 3 and suitable both for patient rehabilitation after injuries or surgeries and for tracking the activities of athletes.

5.1. Devices and Systems for Post-Operative Rehabilitation

Patients' joint movements are often monitored with two useful technologies: inertial sensors (IMU) and vision-based sensors (e.g., Microsoft Kinect) [20,97,99,107,108]. Ianuculescu et al. proposed a device able to provide post-operative tracking of patient rehabilitation from home, achieving the same therapy results reached in a hospitalized structure [54]. The re.flex system consists of two IMU sensors equipped with a three-axis gyroscope and an accelerometer for tracking knee movements. A proper software application for mobile devices was also developed to acquire and process the acquired data from the IMU sensors. Combining the data collected by the accelerometers and the gyroscopes, the system can measure up to twelve parameters, such as speed, gravity, and three-dimensional orientation, allowing the user to monitor movements during exercise training.

Additionally, the mobile application allows the user to interface with the measured data; the patient and the physician can visualize the movements of the leg through a real-time three-dimensional avatar and follow the exercises selected and customized by the physiotherapist. A self-calibrating algorithm automatically detects the sensor's position and calibrates it accordingly, to achieve the best output response. This automatic calibration allows the user to easily attach the sensors to the injured area. The system was tested for monitoring the reconstruction of the anterior cruciate ligament of 30 patients. They were randomly separated into two groups 48 h after the surgery: the first group followed rehabilitation in a clinic, and the second group stayed at home using the re.flex monitoring system. For both groups, the treatment lasted three months. The group treated remotely with the re.flex system achieved an average Oxford knee score—an index that measures knee pain and function after reconstruction surgery—of 1.3 points higher than the group treated in the clinic.

A capacitive sensor with a bio-inspired eggshell microstructure was proposed by He et al. for monitoring walking, air blowing, finger, knee, and elbow movements, and performing voice recognition [29]. The sensor was made of a thin dielectric layer with microstructured PDMS geometries embedded between two electrodes of MXene titanium carbide (Ti_3C_2Tx) mixed with Ag nanowires (Figure 9a). The sensor showed excellent sensitivity (i.e., 10.13 MPa⁻¹) and a good response time (~ms) within a pressure range from a few kPa up to 600 kPa (Figure 9b).







Figure 9. A capacitive pressure sensor's construction (**a**). Illustration of the ESIM's (eggshell inner membrane) structural elements (**i**); schematic showing the construction of MXene sheets (**ii**); and the manufacture of flexible capacitive pressure sensors (**iii**). Schematic sensor diagram detecting human physiological signals and body motions. From top left to bottom right: vocal cord vibration signal when pronouncing the word "studying"; capacitance responses from the repeated air blowing, clenching and releasing of finger/knee/elbow bending, and walking (**b**) [29]. Sensor output during exhalation and inhalation acquired as capacitance variation during deep breathing, measured in different positions: standing, sitting, and lying (**c**). Variation in the capacity during the slight vibrations of the vocal cords allowed to recognize "Nice to meet you", "Capacitance", "Research is my forever love", "The University of Manchester", "Thank you", and "Freedom" (**d**) [53].

Chen et al. developed a capacitive sensor with a textile structure, excellent conductivity and elasticity, and high comfortability when worn [53]. The sensor was fabricated by coating the starting fabric with nickel. A properly shaped fabric, with a size of 13 mm \times 37 mm, was applied to the patient's belly to evaluate breathing during rehabilitation exercises. One

18 of 40

of the two electrodes was inserted inside the tights, the other into the inner part of a belt (Figure 9c).

By inhaling and exhaling, the patient expands and contracts the abdomen, varying the distance between the two electrodes and thus generating a capacitance variation (as shown in Figure 9d). Slight vocal cord vibrations were also monitored to distinguish sentences pronounced by the patient to diagnose and prevent dysphasia issues. After repeated bending cycles, a stress test was performed to verify the signal stability. After hundreds of bending cycles, the device's capacity increased from 24.3 to 26 pF, showing good signal stability even after running a high number of cycles. Similarly, in [109], the authors presented novel Ag–NW-based capacitive strain sensors to sense skin deformations for tracking finger movements. The sensors' characteristics were a high sensitivity (gauge factor \approx 1), low Young's modulus (96 kPa), linear response, high stretchability (150%), negligible hysteresis, and broad strain sensing range greater than required to detect human movements. The experimental results demonstrated that the skin deformations acquired with the developed capacitive sensors are highly correlated with those obtained with an eight-camera optical tracking system. The authors suggest using these sensors to monitor hand movements in patients affected by chronic diseases such as cerebral palsy, spinal cord injury, or muscular dystrophy.

Other hand-worn systems were developed to assist users in rehabilitation from stroke and traumatic brain injury [110]. In Ref. [111], the authors presented an innovative robotic system for assisting patients with repetitive hand exercises. Specifically, they created a wearable therapy device driven by pneumatic muscles (PM) that delivers the supportive forces needed for gripping and finger extension. The robot fingers have two independent degrees of freedom (DoF); also, position and force data are acquired by integrated sensors for controlling the robot and evaluating its performance. A fuzzy PID controller was developed to control the robot's movements, which, as demonstrated by experimental tests, performs better than the standard PID controller.

In Ref. [112], the authors presented a hand rehabilitation system that aids patients in performing repetitive training. It comprises a Myo armband, a robotic glove, a Unity3D, and a videogame processing platform. The Myo armband gathers spatial and gestural data using integrated IMU and EMG sensors. The acquired data are wirelessly transmitted to Unity3D, which derives the controls for the robotic hand. This last device includes two stepper motors for controlling the opening/closing of the hand. In a hand-training game created in Unity 3D, players must pick up, hold, move, and drop a cube in progressively challenging puzzle stages. To make the user feel more engaged in the game, the user sees virtual hands and arms that mimic their actions.

Furthermore, with 11 elastic actuators that impart extension torques to the finger and thumb joints, the HandSOME II device helps the patient expand their hand [113]. Despite the device's DoFs not being aligned with those of human joints, a new mechanical design delivers forces orthogonally to the finger segments. Experimental tests were carried out on seven post-stroke patients affected by hand disabilities, demonstrating that HandSOME II can enhance the extension angles and range of motion in all finger joints.

Innovative materials are continuously developed for implementing sensors to detect body movements and physiological parameters, which can easily be integrated into the clothes. Tan et al. developed a textile-based piezoresistive sole to monitor the patient's walking motion during rehabilitation and prevent diseases such as diabetic ulcers and foot deformation [62]. The sensor was made with a cotton electrode coated with reduced graphene oxide (rGO) and a fabric electrode with a silver (Ag) circuit inside. Six sensors were inserted into the shoe sole, allowing for real-time observation of the foot pressure distribution during movement, achieving an accurate pressure map. The sensors were connected to a small electronic board equipped with an acquisition circuit, a Bluetooth module for wireless data transmission, and a battery for power supply. The sensor had a 170 ms response time, a 3.90 kPa⁻¹ sensitivity in a pressure range between 0 and 36 kPa, and exhibited good stability during a durability test of up to 1000 cycles.

Weak stimuli such as breathing, swallowing, chewing, finger flexion, muscle vibration, and joint movements were successfully measured by exploiting a piezoelectric sensor developed by Wang et al. [21]. The sensor was made of poly(vinylidene fluoride-trifluoro ethylene) multi-walled carbon nanotubes (p(VDF–TrFE)/MWCNTs) grown in their β -phase and sandwiched between two gold electrodes. A transparent thin silicone film covered the device and protected it without altering the final flexibility. The sensor's sensitivity and extensibility were calculated, resulting in 540 mV/N and 0.986 Gpa, respectively. Studies on adult patients were carried out, placing the sensor in contact with the skin on the wrist and acquiring the heartbeat. A measurement of 84 beats per minute-corresponding to a standard heartbeat—was recorded even with a very weak input stimulus. The breathing frequency was also determined with a peak voltage of 0.1 V, as well as the swallowing, by applying the sensor to the throat. A peak voltage of 0.75 V was easily detected, allowing measuring also the time the muscles take to complete the whole swallowing movement. Chewing gestures were finally measured. Here signal peaks of 0.3 V amplitude were detected. Finger bending and bending of the wrist were also detected. Figure 6a shows the plot of the acquired signal during the previously mentioned tests. Kim et al. proposed an innovative sensor to measure gait by exploiting the piezoelectric effect [24].

The sensor, made of a piezoelectric PVDF layer, was fixed on one side by two rubber bands, while on the opposite side was attached a polyester thread where the mechanical tension was applied (Figure 10a). A flexible printed circuit board was used to acquire the output voltage by connecting two electrodes to the upper and lower surfaces of the PVDF film with carbon-based conductive tape. The upper surface's ends of the PVDF strip were protected by a polyethylene terephthalate (PET) layer. Only one of the lower surfaces was fixed to the PET substrate, allowing the PVDF strip to flow freely over the substrate when a traction stimulus was applied to the free side. Kapton tape was used to electrically insulate the sensor. The peak-to-peak open circuit voltage was measured as a function of the upper limb movement; seven sensors were then embedded in commercial stockings, and medical bands were applied to the wrist, elbow, chest, and shoulder.



Figure 10. Cont.



Figure 10. PVDF flexible sensor and the corresponding output voltage over time, and position for the detection of heartbeat, breathing, swallowing, and chewing. Additionally, maximum voltage values depending on the flexion levels of the finger, upper bending, lower bending, and wrist rotation are also reported (**a**) [21]. Flexible piezoelectric sensor with the corresponding exploded drawing showing the multilayered structure (**b**). Here the deglutition wave recognition and segmentation with the comparison of the sEMG recorded signals of a single swallow action are reported. Additionally, spontaneous frequency, duration time, and latency of a swallowing act are also displayed [34].

A minimally invasive sensor for monitoring of the deglutition capabilities of the patients to evaluate dysphagia issues was developed by Natta et al. [34]. A piezoelectric sensor worn directly on the throat was used to transduce larynx movements. The sensor was made using a piezoelectric aluminum nitride (AlN) thin film, sandwiched between two molybdenum (Mo) electrodes deposited on a very thin flexible polyimide substrate, reaching an overall thickness of 26 μ m and a weight <2 g. A sticky PDMS–PEIE polymer layer was then used to attach the sensor to the skin (Figure 10b). The output voltage was successfully transmitted to a smartphone via wireless Bluetooth technology after being amplified and filtered by a suitable conditioning circuit. Duration of the swallowing act, frequency of spontaneous saliva deglutition, and latency were easily detected by this sensor, allowing an objective evaluation of the subject's swallowing capability and providing, at the same time, an early diagnosis of pathological conditions. The data collected were compared with standard surface electromyography (sEMG) signals, showing that the sensor is a valid alternative to classical devices (Figure 10b).

A sensor to detect and monitor swallowing by converting tongue movements into an output voltage generated by the triboelectric effect in single-electrode mode, also exploited to self-power the sensor itself thereby significantly reducing the power consumption of the whole system, was developed by Yun et al. [33].

Three layers of different materials were used for fabricating the triboelectric-nanogenerator for swallowing rehabilitation (TSR): a layer of polytetrafluoroethylene (PTFE) for triboelectric induction, a copper (Cu) electrode, and a top layer of polyethylene terephthalate (PET). The triboelectric nanogenerator for swallowing rehabilitation was then applied to the palate to detect tongue movements. It demonstrated good sensitivity of 47 mV/kPa even to weak tongue pressures ranging from 20 to 100 kPa. Two different TSR devices were then characterized, the TSR–B with bare PTFE and the TSR–T60 with 60° tilted PTFE. Peaks of the open-circuit voltage of approximately 13.10 V (with a short-circuit current of 0.512A) were measured for the TSR–B. An increment of the output voltage of approximately 189%—approximately 24.76 V—was observed for the tilted TSR–T60. The power density was then calculated to define the energy generation for self-power low-consumption devices when the sensors were connected to a variable resistance load.

Additionally, the sensors' sensitivity was calculated by analyzing the output voltage curve as a function of the applied pressure. In this case, the pressure range was split into two sub-ranges, namely R1 from 20 to 180 kPa and R2 from 300 to 3.4 MPa. The voltage vs. pressure curve slope was then 35 mV/kPA for the TSR–B and 47 mV/kPa for the TSR–T60 in the range R1 and 1.10 mV/kPa and 2.20 mV/kPa in the range R2, respectively. Finally, humidity tests were also performed, increasing the humidity percentage from 30% to 80% to characterize the sensor in a humid environment, reproducing similar test conditions to the mouth. Figure 11 displays a schematic illustration of the triboelectric-based sensor following one cycle of tongue movement, the generated output response, the sensitivity, and its performance when changing the humidity and environmental conditions.



Figure 11. Schematic illustration of fabricated TSR and the working mechanism in one operating cycle (**a**). Open circuit voltage and short circuit current measured on TSR–B and TSR–T60, including the tension produced when the tilting varies, the pressure sensitivity of both the sensors, the output voltage when the humidity increases up to 80%, and the power density according to the load resistors (**b**) [33].

Besides post-operative rehabilitation after trauma or injuries, numerous wearable devices were presented in the scientific literature for rehabilitation after damage to the nervous system due to stroke. In particular, G. Li et al. presented an untethered adaptable thumb exoskeleton that actively supports the thumb's three degrees of freedom of motion [114]. An adaptive thumb and spherical mechanisms constitute the exoskeleton, including sliders made of ball-type linear bearings to reduce the rolling friction. The exoskeleton's workspace, self-alignment, interface forces, admission controller, and grasping aid performance were tested experimentally. The workspace and self-aligning tests indicated that the exoskeleton could obtain a significant thumb joint workspace and self-alignment. The test results suggested that the presented robot-assisted rehabilitation system could reduce the tangential interaction force by 76.8% and enhance comfort. Additionally, experiments demonstrated that the exoskeleton could improve the thumb's grasping capability.

Similarly, in [115], the authors used wearable inertial sensors placed on different body parts, which gather the movement data and transmit them through Zigbee to a PC. Pre-processing methods were applied to enhance the signal quality before extracting movement features through the multi-sensor fusion technique. Then, the DTW (dynamic time warping) method was employed to determine movement scores. These scores are provided as input to a machine learning model (SVR—support vector regression), using the FMA (Fungl–Meyer assessment) indexes as the output classes' labels. The trial research involved 120 stroke patients, and statistical evaluations of the two assessment techniques were conducted. The experimental results revealed that R² for the score regression analysis for both approaches was 0.9667, whereas the mean deviation was 0.30., The percentage of deviations and relative deviations inside the interval mean \pm 1.96 SD (standard deviation) were equal to 92.50% and 95.83%. Additionally, Y. Ren et al. proposed a robotic ankle device to generate extensive passive and active movement training in post-stroke patients [116]. It includes a force sensor, a rotational actuator, a leg brace, and a foot holder. Adjusting the straps may accommodate each person's leg and foot size. The device is actuated by a DC motor controlled by an encoder to measure the angle change. The resulting system guides the patients in motor relearning, providing motivational feedback to the patient during the rehabilitation training in the active training modality. Experimental tests on ten post-stroke patients demonstrated that 12 sessions could improve motor control ability and neuroplasticity.

Table 4 outlines some of the most important specifications of the analyzed sensors and devices so far, performing a classification according to sensing technology, used materials, sensitivity, transmission technology, area of the body where the sensor is usually applied, response time, and operating range.

Reference	Sensing Technology	Materials	Sensitivity	Range of Detection	Body Area of Application	Measured Parameters	Response Time	Transmission Technology
Ianculescu et al. [54]	IMU	N.A. (*)	N.A. ^(*)	N.A. (*)	Knee	Knee bending angle	N.A. ^(*)	Wireless
He et al. [29]	Capacitive	MXene (Ti ₃ C ₂ Tx)/AgNWs	0.01–2.04 kPa ^{-1 (A)}	0–2 kPa	Throat, cheek, finger, knee, elbow, foot	Voice recognition, air-blowing intensity, joint bending angle, walking frequency, and weight pressure	ms	Wired
Chen et al. [53]	Capacitive	Nickel-coated fabric	N.A. ^(*)	N.A. ^(*)	Belly, throat, finger	Breathing frequency and intensity, voice recognition, finger bending angle	5–50 µs	Wired
Lei et al. [28]	Capacitive	ACC/PAA/alginate hydrogel	0.17 kPa ^{-1 (A)}	0–1 kPa	Finger, throat, wrist	Finger bending angle, voice recogni- tion, heartbeat	N.A. ^(*)	Wired

 Table 4. Specification comparison of analyzed rehabilitation monitoring systems.

Reference	Sensing Technology	Materials	Sensitivity	Range of Detection	Body Area of Application	Measured Parameters	Response Time	Transmission Technology
Park et al. [61]	Capacitive	PDMS AgNW CFs	0.015 kPa ^{-1 (A)}	10–50 kPa	Belly	Breathing frequency and intensity	N.A. ^(*)	Wired
Yao e Zhu [55]	Capacitive	AgNW/PDMS Ecoflex	1.62 MPa ^{-1 (A)} below 500 kPa	Up to 1.2 MPa	Knee, finger	Finger bending and knee bending frequency	40 ms	Wired
Sheng et al. [57]	Capacitive	GaInSn	N.A. ^(*)	Stretchability up to 250%	Elbow, wrist	Wrist and elbow bend angles	<10 ms	Wireless
Yun et al. [33]	Triboelectric	PET/Cu/PTFE	47 mV/kPa	20–100 kPa	Tongue	Pressure and frequency of tongue movements	N.A. ^(*)	Bluetooth
Tan et al. [62]	Piezoresistive	rGO-Ag	3.90 kPa ^{-1 (B)}	0–100 kPa	Foot	Pressure distribution on the foot sole	170 ms	Wired
Tognetti et al. [25]	Piezoresistive	Carbon-coated PA and lycra elastic yarns	(Angular sensitivity) 960 Ω/° ^(C)	N.A. ^(*)	Knee	Knee bending angle	N.A.	Wired
Ge et al. [56]	Piezoresistive	PDMS AgNW	$4.29 \text{ N}^{-1 (D)}$	0–2 N	Wrist, finger	Heartbeat and finger bending angle	8 ms	Wired
Zhu et al. [32]	Piezoresistive	TPU/CNT-CNC	GF = 321	Stretchability > 500%	Throat, finger, elbow, knee, neck, chest	Voice recognition and finger, elbow, knee, and neck bending degree	N.A. ^(*)	Wired
Dan et al. [30]	Piezoresistive	PDMS/AgNW	14.1 kPa ⁻¹ , 4.8 kPa ⁻¹ , 1.84 kPa ^{-1 (B)}	(0–3.5) kPa, (3.5–10) kPa (10–40) kPa	Wrist, nose, cheek, throat	Heartbeat, exhalation frequency, facial expression signals, and voice recognition	47 ms	Wired
Kim et al. [27]	Piezoresistive	PUD/CNTs	0.31 kPa ⁻¹ , 0.1 kPa ⁻¹ , 0.03 kPa ^{-1 (E)}	<1000 Pa, (1–20) kPa, >20 kPa	Wrist, throat, finger, cheek	Arterial and jugular heartbeat, finger bending angle, cheek bulging frequency, voice recognition	36.7 ms	Wired
Lu et al. [44]	Piezoresistive	PANI/PAAMPSA	GF = 1.7 (100% strain) GF = 14.52 (1500% strain)	Stretchability up to 1935%	Wrist, elbow, finger, ankle, knee	Finger, elbow, wrist, and knee bending angle, frequency of walking and jumping	N.A. ^(*)	Wired
Kim et al. [24]	Piezoelectric	PVDF-elastic threads	N.A. ^(*)	0–5 N	Shoulder, forearm, elbow, wrist	Bending and rotation angle of shoulder, wrist, and forearm	N.A. ^(*)	Bluetooth
Wang et al. [21]	Piezoelectric	(P(VDF– TrFE)/MWCNT)	540 mV/N	0.5–5.0 N	Wrist, throat, cheek, nose	Heartbeat, intensity and frequency of breathing, swallowing and chewing	N.A. ^(*)	Wired
Natta et al. [34]	Piezoelectric	AlN/Mo on Kapton	0.025 V/N	10–50 kPa	Throat	Frequency, duration, and latency of swallowing	15 ms	Wireless

Table 4. Cont.

(*) N.A. Not Available. ^(A) Sensitivity is calculated by normalizing the capacitance variation as $\frac{\Delta C_{C_0}}{P}$ where P is the applied pressure. ^(B) Sensitivity is calculated by normalizing the resistance variation as $\frac{\Delta R_{C_0}}{P}$ where P is the applied pressure. ^(C) Angular sensitivity is calculated as $\Delta R_{\Delta \alpha}$ where α is the bending angle. ^(D) Sensitivity is calculated by normalizing the resistance variation as $\frac{\delta(\Delta R_{R_0})}{\delta(F)}$ where F is the applied force. ^(E) Sensitivity is calculated by normalizing the current variation as $\frac{\delta(\Delta R_{R_0})}{\delta(F)}$ where P is the applied pressure.

5.2. Devices and Systems for Tracking an Athlete's Performance

The current wearable sensors for sports tracking still have several limits to overcome, such as the high cost and high energy consumption, as well as poor fitting features and elasticity making them uncomfortable to be worn and hamper the movements during sports activities. However, innovative and challenging solutions have recently been developed in

different sports fields, such as combat sports, basketball, volleyball, soccer, running, and weightlifting [117–124].

Masihi et al. designed a capacitive pressure sensor to detect impacts on the player's head while using a helmet [46]. The transducer consists of two conductive textile electrodes and a porous layer of PDMS as a dielectric layer. The porosity of the dielectric layer was obtained by dissolving carbon dioxide (CO₂) gas in a compound of PDMS and sodium bicarbonate (NaHCO₃) and mixing it with nitric acid (HNO₃). With this mixture, nine sensors with different porosities and dielectric constants were fabricated. The difference in the dielectric constant is mainly due to the percentage of HNO₃ used. From laboratory tests, the device with 15:1 as the PDMS/HNO₃ ratio had the highest sensitivity of 0.3 kPa⁻¹ for applied pressures <50 Pa and 3.2 MPa⁻¹ in the pressure range from 0.2 to 1 MPa.

De Fazio et al. developed a self-powered smart insole based on piezoresistive and inertial sensors (Figure 12a). The insole relies on a matrix of eight custom piezoresistive sensors developed using a pressure-sensitive polymeric layer (i.e., Velostat), whose resistive values are read by an acquisition system made with a multiplexer and Arduino Lilypad [63]. The insole is also equipped with an inertial sensor to monitor walking and step counting, a power module with a battery, and a piezoelectric harvester to recharge it, exploiting the same pressures applied during gait and walking movements. A Bluetooth module is used to transmit the data to a PC. The insole was tested while walking at a speed of 1 m/s, and the acquired data were used to build a distribution pressure map.

A hybrid sensor used to monitor the technical movements of taekwondo athletes was made by Ma et al. The proposed e-textile sensor can detect elongation and pressure stimuli by exploiting both piezoresistive and capacitive effects [40]. The sensor was fabricated starting from commercial polyacrylonitrile (PAN) fibers, coated with carbon nanotubes (CNTs), and reduced graphene oxide (rGO) to make it conductive. Then, the two achieved fibers were rolled up, forming a helical structure wound around a polyurethane (PU) multifilament, obtaining a conductive core-sheath yarn sensitive to stretch (Figure 12b).

The sensor was designed considering the dobok taekwondo uniform should withstand substantial pressure impacts (>100 kPa) and extensive stretch movements (with a strain > 50%). Core-sheath yarns were fabricated with an increasing winding number from 20 to 70 along a distance of 10 cm. To test the elongation ability of the e-textile when used as a strain sensor, it was stretched, and the relative resistance variation $\Delta R/R_0$ as a function of the stretching was measured. The characterization results showed that yarns from 20 to 50 twists exhibited a negative variation in the resistance response, while yarns with 60 and 70 twists positively increased. Experiments were also performed to characterize the capacitive behavior as a pressure sensor by acquiring the relative capacity variation $\Delta C/C_0$ in a range of pressures up to 110 kPa (a typical practical case). In the ranges 0–9 kPa, 9–37 kPa, and 37–110 kPa the sensitivity was then calculated, and values of 0.112 kPa⁻¹, 0.0283 kPa⁻¹, and 0.0021 kPa⁻¹, respectively, were reported. The sensor also had a fast response to pressure pulses between 1.5 and 10 kPa and good stability of the output response even after 100,000 cycles of long tests. Moreover, practical cases were successfully analyzed, and the sensor was integrated into the uniform on the chest to identify the pressure distribution of a blow and on the knee to monitor the athlete's kicks, allowing to identify the angle of the bending knee (the experimental results are reported in Figure 12b).



Figure 12. Examples of capacitive and piezoresistive hybrid sensors: (**a**) The developed sole with the piezoelectric integrated sensors, the acquisition system, and the Bluetooth module. Below, the characteristic curves R vs. F (in Kgf and with a logarithmic scale) for five Velostat-based pressure sensors are shown for different sizes, $3 \text{ cm} \times 3 \text{ cm}$, $1 \text{ cm} \times 1 \text{ cm}$, and $3 \text{ cm} \times 1 \text{ cm}$ [63]. The yarn matrix structure in the inset shows some examples of the final fabric's elongation, bending, and twisting (**b**). A map of the pressure distribution during the hit cycle and resistance variation at different knee bending angles are shown on the right. Some practical scenarios of using the integrated sensor on the chest and knee are also reported [40].

A piezoelectric pressure sensor for a wireless wearable sensing system obtained from the combination of tetragonal-phase BaTiO₃ nanowires and electrospun PVDF nanofibers was designed and proposed by Guo et al. [22]. The system also includes an electronic board with a module for amplifying the acquired signal, a power converter, a module for controlling the data conversion, and a Bluetooth module for wireless transmission. The output voltage was evaluated using an ad hoc set-up and applying a controlled pressure. The maximum output current value measured on 100% PVDF fibers was approximately 47 nA, while that measured on the PVDF nanocomposite with 3% BaTiO₃ nanowires was 105 nA. The sensor exhibits a sensitivity of 0.017 kPa⁻¹ and a response time of 290 ms, estimated from the voltage-over-time curve in one contact cycle. A durability test of 1750 contact cycles at a frequency of 3.5 Hz was also performed, showing the sensor has very good signal stability over time. Practical tests were carried out. The sensor was then embedded inside a sole to track walking, running, and squatting movements. Current peaks of 97 nA, 241 nA, and 331 nA were measured, respectively. The sensor was also attached to the elbow, evaluating the flexion and extension angles at a frequency of 1.15 Hz. Here, the maximum current generated was 36 nA at an elbow bending angle of 120°. The sensor was finally used to detect slight vibrations of the vocal cords when placed when pronouncing short sentences, such as "Hi" and "Oh my God". The acquired signal during these tests is reported in Figure 13a.

Still exploiting the piezoelectric effect, Zhao et al. designed a self-powered sensor to monitor basketball players' movements and gestures [39]. The sensor was fabricated starting from a mixture of dimethyl formaldehyde (DMF) and PVDF poured onto a silicone substrate. Then, thin Ag electrodes of 300 nm thickness were applied on both sides of the piezoelectric film. After wiring it, the sensor was attached to the elbow. Generated voltage spikes of 2.172 V at an angle of 150°, 3.48 V at 120°, 6.052 V at 90°, and 8.08 V at 60° were measured. Moreover, the relative piezoelectric output at different frequencies was also acquired, keeping the same bending angle. Voltages of 5.24 V, 5.24 V, 5.328 V, and 5.32 V were measured at 0.5, 1, 1.5, and 2 Hz, respectively. Finally, the sensor was positioned on the popliteal fossa of an athlete for in vivo tests; the generated voltage during the jumping, walking, and running activities was successfully acquired, showing voltage peaks of 9.387 V, 1.02 V, and 2.04 V, respectively (Figure 13b).

Liu et al. proposed a similar self-powered piezoelectric sensor composed of polarized PVDF film for voice recognition and monitoring of arm and hand movements of volleyball players. Here, the sensor was also used as an energy harvester to scavenge voltage from the continuous movements of the volleyball player to charge a capacitor and power up a Bluetooth module for wireless transmissions [42]. The signal generated by the sensor during hand bending and movements is displayed in Figure 13c.

In addition, Li et al. developed a pressure sensor that, thanks to its high sensitivity (approximately 1.9 V/kPa^{-1}), was used to detect very slight athlete movements and prevent overtraining injuries [64]. The sensor was made of polyvinylidene fluoride and hexafluoro-propylene (PVDF–HFP) nanofibers, obtained through electrospinning and painted with zinc oxide (ZnO) solution using an atomizing gun. The piezoelectric layer obtained was sandwiched between two aluminum layers. Wires were then used to collect the voltage generated by the sensor. Different amplitude forces—from 0.02 N up to 0.5 N—at a frequency of 1 Hz were applied, and the corresponding response time was estimated at approximately 20 ms. Furthermore, the sensor's sensitivity was calculated, with a maximum value of 1.92 V/kPa^{-1} . Activities such as running and breaking, usually performed continuously during a match, were monitored (Figure 13d).



Figure 13. Piezoelectric sensors: PVDF/BaTiO₃-based sensor integrated into a sole with the corresponding output currents generated during squatting, walking, and running activities, elbow extension and flexion to 60°, 90°, and 120°, and the pronunciation of short sentences (**a**) [22]. PVDF/DMF-based sensor applied on an athlete's elbow and the corresponding voltage signals generated at different bending angles and during different physical activities (**b**) [39]. A schematic diagram of different bending angles of the palm during the test of the PVDF sensor. The output piezoelectric voltages of three subjects when finger and elbow bending angle change are also reported (**c**) [42]. Soccer player motion monitoring test using the PVDF–HFP-based sensor (**d**). Pictures of a soccer players' actions, showing normal ankle motion, abnormal ankle motion, normal knee motion, and abnormal knee motion. Moreover, the "brake" action of the same motion is also monitored and reported [64].

Li et al. [41] developed a low-cost, sweat-resistant triboelectric (BSRW–TENG) sensor for tracking movements during individual exercises such as leg and bicep curls and running. The sensor was used to monitor and track three different sports activities, including tests before and after sweating, to verify the waterproofing ability of the materials used and the stability of the output response. In the first case, the sensor was applied to the elbow to monitor the degree of bending during the biceps curl, obtaining feedback on the correct execution of the exercise. Similarly, the sensor was fixed on the back of the knee to monitor



the leg curl. Finally, it was embedded into the sole of the shoes to detect the steps during walking and running. The acquired signals are displayed in Figure 14.

Figure 14. BSRW–TENG sensor applied to the elbow for monitoring biceps curl, leg curl, running, and walking—before and after sweating—with the corresponding plots of generated output voltages [41].

Table 5 outlines the most important specifications of the analyzed systems for tracking sports activities, classified according to sensing technology, used materials, sensitivity, transmission technology, area of the body where the sensor is usually applied, and response time and transmission technology used.

In Figure 15, a radar chart is reported, comparing the analyzed sensors and devices, both for rehabilitation and sports tracking purposes, according to their technological maturity, flexibility, ability to be completely integrated into clothes and garments, sensitivity, and range of detection.

Reference	Sensing Technology	Materials	Sensitivity	Range of Detection	Body Area of Application	Measured Parameters	Response Time	Transmission Technology
Masihi et al. [46]	Capacitive	PDMS	0.3 kPa ⁻¹ 3.2 MPa ^{-1 (A)}	<50 Pa 0.2–1 MPa	Head	Head pressure distribution	115 ms	Wired
Li et al. [41]	Triboelectric	PDMS-elastic resin	N.A. ^(*)	2–260 N	Elbow, knee, foot	Knee and elbow bending angles, pressure, and frequency of steps in running and walking	N.A. ^(*)	Wired
Yang et al. [50]	Triboelectric	TPU/silicone rub- ber/conductive fabric	0.054 V/kPa ⁻¹	2–200 kPa	Finger, wrist, elbow, foot	Finger, wrist, and elbow bending angles; plantar pressure distribution during walking, running tiptoe, and jumping	N.A. ^(*)	Wired
Guo et al. [22]	Piezoelectric	PVDF/BaTiO ₃ (NW)	0.017 kPa ^{-1 (B)}	1–40 kPa	Foot, elbow, throat	Elbow bending angle, voice recognition, pressure, and frequency of steps in running and walking	290 ms	Bluetooth
Zhao et al. [39]	Piezoelectric	PVDF/Ag/PET	N.A. ^(*)	N.A. (*)	Elbow, knee, wrist, finger, throat	Vocal recognition; elbow, wrist, finger, and knee bending angle; walking, jumping, and running frequency	N.A. ^(*)	Bluetooth
Li et al. [64]	Piezoelectric	(PVDF- HFP)/ZnO	1.92 V/kPa ⁻¹	0.02–0.5 N	Knee, ankle	Frequency and degree of the knee and ankle bending during running	20 ms	Bluetooth
Liu et al. [42]	Piezoelectric	PVDF	N.A. ^(*)	N.A. ^(*)	Throat, elbow, finger	Vocal recognition, finger and elbow bending angles	N.A. ^(*)	Wireless
Saponara et al. [47]	IMU, strain gauge, electro- goniometer	Aluminum	N.A. ^(*)	Hundreds of g (gravity force)	Hip, knee, elbow	Speed, acceleration, pressure, trajectory, and response time of punch and kick	N.A. ^(*)	Bluetooth
Ma et al. [40]	Resistive- capacitive	GO-CNT/PU e-textile	0.1124 kPa ⁻¹ , 0.0283 kPa ⁻¹ , 0.0021 kPa ^{-1 (A)}	0–9 kPa, 9–37 kPa, 37–110 kPa	Chest, knee	Hit pressure distribution and knee bending angles	120 ms	Wired
Zhu et al. [18]	Triboelectric/ piezoresistive	PTFE/latex– PVDF/hydrogel	N.A. ^(*)	N.A. ^(*)	Wrist	Wrist bending and rotation angles	N.A. ^(*)	Bluetooth
Mariello et al. [43]	Triboelectric/ piezoelectric	PDMS/ Ecoflex- AIN/Mo	59.4 mV/kPa ⁻¹ 160 mV/kPa ⁻¹ , 3.7 mV/kPa ⁻¹	0–50 kPa, 50–120 kPa, 120–400 kPa	Foot, elbow, wrist, finger, ankle, knee, neck	Hit pressure on human skin; walking and running speed; finger gestures; ankle, elbow, neck, wrist, and knee bending	N.A. ^(*)	Wired

Table 5. Specification comparison of analyzed sports activity monitoring systems.

(*) Not Available. ^(A) Sensitivity is calculated by normalizing the capacitance variation as $\frac{\Delta C/C_0}{P}$ where P is the applied pressure. ^(B) Sensitivity is calculated by normalizing the voltage variation as $\frac{\Delta V/V_s}{dP}$ where P is the applied pressure and V_s is the saturation voltage.

The reported Figure 15 integrates evaluation criteria that are difficult to objectively estimate, referring to miscellaneous and very different properties for each analyzed technology. In particular, flexibility is evaluated as the maximum percentage of the analyzed work proposing fully flexible devices. Integrability is, instead, derived considering the number of analyzed works proposing a technology fully integrated or potentially integrable. Here, 10 (the maximum) is assigned to devices already integrated, whereas 5 and 0 are used for devices potentially integrable and not integrated at all, respectively.

Finally, technological maturity is determined considering the number of works/papers proposing that transduction mechanism so that the maximum number of cited works corresponds to 10 in the chart of technological maturity. These criteria have been qualitatively graded based on the authors' judgment following a thorough review of the recent available literature. They represent a dynamic picture of the current state-of-the-art related to the analyzed applications. Despite being qualitative, the authors think the set of criteria is important for the scientific community to evaluate the advantages and disadvantages of each technology. Therefore, larger-surfaced hexagons should, in theory, represent the best transduction mechanisms to be utilized.



Figure 15. Radar chart showing a comparison between the analyzed sensors and devices according to their technological maturity, flexibility, ability for full integration into clothes and garments, sensitivity, and range of detection.

Finally, from these radar graphs, it is clear that piezoelectric sensor technology is the most mature, considering the numerous advances made in recent years from the point of view of performances (i.e., sensitivity, stability, robustness), integrability, manufacturing process, and costs. Additionally, piezoresistive and capacitive sensors offer good performance in terms of sensitivity; thanks to textile-based and polymeric solutions, such sensors enable their simple integration into cloths. However, piezoresistive sensors have a relatively reduced detection range, limiting the applicability of this sensor typology for detecting large movements. Triboelectric sensors allow high flexibility and integrability thanks to the multitude of materials that can be used for realizing such devices, including polymeric materials.

5.3. Overview of Datasets Related to Sport and Rehabilitation Applications

In the last few decades, the scientific community and companies have created several datasets containing heterogeneous signal parameters (e.g., accelerations, angles, pressures, etc.) that were acquired through different approaches (e.g., inertial, optical, piezore-sistive, video, etc.), making them publicly available to researchers for developing future applications for rehabilitation and sports performance monitoring.

Considering the datasets gathered using camera-based and optical systems, the UI–PRMD (University of Idaho–Physical Rehabilitation Movement Data Set) collects motions associated with typical exercises performed by patients in physical therapy and rehabilitation programs [125]. The UI–PRMD is a freely accessible dataset, including ten rehabilitation movements; ten healthy people performed each motion ten times in front of Vicon optical trackers and Kinect cameras. The acquired data are the locations and angular orientations of the body joints in the skeletal models measured using both MoCap (motion capturing) systems. Likewise, the KIMORE (KInematic assessment of MOvement and clinical scores for remote monitoring of physical REhabilitation) dataset is a free dataset containing different rehabilitation exercises collected by an RGB–D sensor [126]. The dataset comprises skeletal data from five workouts performed by seventy-eight participants; these data are a time series of skeleton joint locations recovered from images taken with a Kinect. Furthermore, in [127], a multimodal dataset was proposed to analyze and measure the quality of movements carried out during karate moves. In particular, an optical motion capture system, consisting of nine cameras, was employed to record the user motions gathered from seven participants. The data underwent post-processing, which included identifying the markers, creating the models, and removing noise caused by "ghost" or jitter markers. The authors used the created dataset to assess the quality of karate moves.

Additionally, several datasets reporting inertial data are available, enabling, for instance, the development of machine learning applications. A daily and sports activities data set" comprises motion data of 19 daily and sports activities (e.g., sitting, standing, lying on the back and right, ascending and descending stairs, etc.) performed by eight subjects for 5 min [128]. In particular, five sensors were applied to five body areas (i.e., torso, right arm, left arm, right leg, left leg), recording inertial data on nine DoF (i.e., x, y, z accelerometers; x, y, z gyroscopes; x, y, z magnetometers). Similarly, in [129], researchers presented the human activity recognition using smartphones data set; the trials were conducted on 30 participants aged 19 to 48. Each participant used a Samsung Galaxy S II smartphone while engaging in six different activities (walking, walking upstairs, walking downstairs, sitting, standing, and lying). Using the phones' integrated accelerometer and gyroscope, they recorded three-axial linear acceleration and three-axial angular velocity at a constant rate of 50 Hz. The resulting dataset was divided into two sets at random, with 30% of the volunteers chosen to create test data and 70% of the participants chosen to create training data.

Moreover, datasets including heterogeneous data typologies are available; for instance, MoVi (motion video) is a publicly available dataset, including video and inertial information [130]. In particular, 60 female and 30 male patients were involved, collecting data during 20 everyday and sports movements. The movements were captured throughout five rounds utilizing an optical motion capture system, video cameras, and inertial measurement units (IMU). The dataset includes 6.6 h of IMU data, 9 h of motion capture data, and 17 h of video from four distinct points of view.

With regard to gait monitoring applications, examples of publicly available datasets are the gait in aging and disease database [131], the MIT database [132], and the Georgia Tech dataset [133]. The first one comprises time series of the walking stride interval acquired from fifteen subjects (i.e., five healthy young adults, five healthy old adults, and five older adults with Parkinson's disease) [131]. On flat ground, the subjects constantly traversed a path free of obstructions. The stride interval was measured using compact force-sensitive resistors inserted into the shoe. An ankle-worn microcomputer was used to sample the analog force signal at 300 Hz using a 12-bit A/D converter while simultaneously recording the data. The interval between foot strikes was then automatically calculated. The datasets discussed above are summarized in Table 6, classifying them from the point of view of the number of participants, the detected parameters, the employed acquisition system, and the suggested application.

Dataset	Provided by	No of Participants	Parameters	Approach	MoCap System Details	Suggested Application
UI–PRMD [125]	University of Idaho	10	Locations and angular orientations of the body joints	Vision-based	Vicon optical trackers Kinect cameras	Monitoring rehabilitation exercises
KIMORE [126]	Marche Polytechnic University	78	Joint locations	Vision-based	Kinect cameras	Detection motor dysfunction
M. Capecci et al. [127]	Marche Polytechnic University	7	Joint locations	Vision-based	Kinect v1	Evaluation of karate moves
Daily and sports activities data set [128]	Bilkent University	8	Inertial data	Sensor-based	Inertial sensors (25 Hz sampling frequency)	Activity recognition
Human Activity recognition using smartphones data set [129]	University of Genoa	30	Inertial data	Sensor-based	Smartphone (Samsung Galaxy S II)	Activity recognition
MoVi dataset [130]	York University	90	Camera images, joint locations, inertial data	Vision-based Sensor-based	15 cameras (Qualisys Oqus 300 and 310) 2 stationary cameras (RGB Grasshopper2) 2 hand-held cameras (iPhone 7) 17 IMU sensors (Noitom Neuron Edition V2)	Motion recognition
Gait in aging and disease database [131]	PhysioBank	15	Stride interval	Sensor-based	Force-sensitive resistors	Normal gait and Parkinson's disease analysis
MIT database [132]	MIT	24	View, time	Vision-based	Sony Handycam	Gait recognition
Georgia Tech [133]	Georgia Tech	20	View, time, distance	Vision-based	-	Gait recognition

Table 6. Table summarizing the datasets previously discussed, classified according to the number of participants, the acquisition system, the detected parameters, and the suggested application.

6. Conclusions and Future Developments

Wearable sensors are a very promising technology for post-rehabilitation and sportstracking applications. The assessment of a patient's physical state, the degree of rehabilitation training, and the quality of the rehabilitation effects are crucial in the recovery process where the collection of a series of data (e.g., movements, bending, joint rotation, and so on) is useful to provide important tracking feedback to the user. These data allow for defining the correct therapies and evaluating their effectiveness and patient progress based on suitable protocols that ensure the patients carry out their physiotherapy programs even at home. Moreover, they allow for optimizing the exercises and assessing the athlete's progress during sports training.

This paper reviewed the state of research on wearable sensors and devices based on different transduction mechanisms focusing on devices at a micro-electromechanical scale. The paper provided an overview of capacitive, piezoresistive, triboelectric, piezoelectric, and inertial transduction methods, distinguishing between wearable sensors for post-operative rehabilitation monitoring and wearable sensors for athletes' performance monitoring. A comparison of the technologies analyzed was then provided, highlighting the biophysical parameters they can detect and some key features that allow defining the performance of the sensors. We believe that our review paper shows several novelties compared to similar review works reported in the scientific literature. In detail, our review work focuses mainly on biomechanical sensors for monitoring joint movements, lingering on aspects related to their design, fabrication, and characterization [107]. Other review works analyze systems for rehabilitation monitoring by considering the system architecture rather than the sensors used [107]. Additionally, we focused on biomechanical sensors (i.e., piezoresistive, piezoelectric, inertial sensors, etc.) for monitoring joint movements, leaving out other sensor typologies (electrochemical sensors, biosignal detection stages, etc.) [10,107–109]. Furthermore, the presented work analyzes sensing systems for detecting both rehabilitation parameters and sports performances; in contrast, other papers deal exclusively with systems for rehabilitation monitoring or sports performance analysis [9,11,107,108]. Furthermore, similar papers discuss wearable sensing systems and other typologies of sensing devices (ambient sensors, portable devices, etc.) or report a general discussion on applying IoT technologies for monitoring sports performance, losing specificity for the considered topic [9,10,107–109].

In addition, our review work explores different sensor technologies in-depth (e.g., piezoelectric, piezoresistive, triboelectric) for joint monitoring without dwelling on specific sensor classes (such as inertial sensors) or specific application areas (e.g., knee) [9,11,108,109]. Finally, comparative analyses are presented in our review for each discussed topic, comparing the performance of the discussed devices and providing insights to establish the systems ranked by best performances. This analysis is a fundamental contribution to our review work, not always shown in other similar works [10,11,107–109]. Table 7 summarizes the advantages of our review paper compared to similar review works. As evident, the proposed review presents a more complete and in-depth analysis of the leading technologies for monitoring rehabilitation and sports performances, not limiting the discussion to specific sensor categories, applications, or monitored body areas.

Table 7. Comparison between the presented review work and similar ones reported in the scientific literature.

References	Limitations of Similar Review Papers	Advantages of our Review Paper
L. do Nascimento et al. [134]	 It mainly deals with the architectural aspects of the discussed systems. It does not consider sensors for monitoring athletic gestures. It deals with other sensor typologies (electrochemical sensors, biosignal detecting stages, etc.). It does not present any comparative analyses. 	 It deals with the design, fabrication, and characterization of sensors for monitoring rehabilitation and sports performance. It considers both systems for monitoring rehabilitation and sports performance. It focuses only on wearable sensing systems. It presents comparative analyses of the discussed systems.
S. Patel et al. [135]	 It does not consider sensors for monitoring athletic gestures. It is not focused on wearable systems. It deals with other sensor typologies (electrochemical sensors, biosignal detecting stages, etc.). It does not present any comparative analyses. 	 It deals with the design, fabrication, and characterization of sensors for monitoring rehabilitation and sports performances. It focuses only on wearable sensing systems. It considers only biomechanical sensors. It presents comparative analyses of the discussed systems.
R. T. Li et al. [136]	 It considers only sensing systems for monitoring sports performances. It mainly deals with motion-tracking systems based on inertial sensors. It does not present any comparative analyses. 	 It deals with sensing systems for monitoring both rehabilitation and sports performance. It focuses on the main typologies of biomechanical sensors for monitoring rehabilitation and sports performances. It presents comparative analyses of the discussed systems.
D. R. Seshadri et al. [10]	 It is not focused on sensing systems for detecting body movements. It considers only sensing systems for monitoring sports performance. It does not present any comparative analyses. 	 It considers only sensing systems for detecting body movements. It deals with sensing systems for monitoring both rehabilitation and sports performance. It presents comparative analyses of the discussed systems.
Y. Zhao and Y. You [9]	 It considers only sensing systems for monitoring sports performance. It reports a general discussion on sensor technology for motion detection, not dwelling on specific systems reported in the literature. It analyses IoT-based wearable systems for monitoring sports performance, not focusing on the sensory aspects. It mainly deals with motion-tracking systems based on inertial sensors. 	 It deals with sensing systems for monitoring both rehabilitation and sports performance. It provides an in-depth discussion of several sensing systems for motion detection reported in the scientific literature. It deals with the design, fabrication, and characterization of sensors for monitoring rehabilitation and sports performance. It focuses on the main typologies of biomechanical sensors for monitoring rehabilitation and sports performance.
S. Bahadori et al. [11]	 It focuses on knee-motion detection systems. It mainly deals with motion-tracking systems based on inertial sensors. It does not present any comparative analyses. 	 It considers sensing systems for monitoring different joints and body movements. It focuses on the main typologies of biomechanical sensors for monitoring rehabilitation and sports performance. It presents comparative analyses of the discussed systems.

With regard to the future perspectives for wearable systems for monitoring rehabilitation and sports performances, several factors must be considered. In addition to the full integration of sensors/actuators, energy sources, procession, and communication within the clothes, future developments of wearable medical systems are facing toward extending the monitoring capabilities, improving the user's comfort, and making the manufacturing process more simple, scalable, and sustainable. In detail, new textile-based sensor structures could be developed, featuring multiple sensing capabilities, reduced invasiveness, and high reliability. Examples of devices complying with such characteristics are already under development, combining different sensing mechanisms into a textile sensor, providing them with extended sensing capabilities and thus improving their applicability [137,138]. Furthermore, advanced electronic textile solutions must be designed, enabling their integration into fiber-based textile devices, and ensuring improved optical and mechanical capabilities. Such solutions must be characterized by low power consumption or have self-sustainable device features, along with being bio-compatible to avoid irritations caused by their prolongated contact with the skin.

Additionally, new haptic feedback technologies could be developed, including inflatable interfaces and dynamic textile shapes (e.g., moving origami textile structures [139]). These solutions may improve the wearing experience of the wearable device, making it more pleasant during long periods of rehabilitation. In this context, customized design and digital manufacturing may be obtained through 3D scanning and 3D printing technologies, altering their shape and functionality depending on specific demands [140]. The possibility of precisely and comfortably positioning the sensors for specific patients is made possible by customization in design.

Furthermore, textile-based displays for discretely providing visual feedback could be developed and integrated into clothes. Recently, textile-based OLEDs were proposed and their usage in practical situations demonstrated [141]. The textile-based OLEDs demonstrated a consistent lifespan under ambient settings, sufficient mechanical toughness to withstand deformation caused by human movement, and washability, i.e., the ability to keep their optoelectronic capabilities even in wet situations such as rain, sweat, or washing.

Author Contributions: Conceptualization, R.D.F., V.M.M. and P.V.; investigation, R.D.F., V.M.M., M.D.V. and P.V.; resources, R.D.F. and M.D.V.; data curation, R.D.F., V.M.M. and P.V.; writing—original draft preparation, R.D.F., V.M.M. and P.V.; writing—review and editing, R.D.F., V.M.M. and M.D.V.; visualization, R.D.F. and P.V.; supervision, M.D.V. and P.V. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Informed consent was obtained from all subjects involved in the study.

Data Availability Statement: Data of our study are available upon request.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Seshadri, D.R.; Li, R.T.; Voos, J.E.; Rowbottom, J.R.; Alfes, C.M.; Zorman, C.A.; Drummond, C.K. Wearable Sensors for Monitoring the Physiological and Biochemical Profile of the Athlete. *Npj Digit. Med.* **2019**, *2*, 114. [CrossRef]
- Seshadri, D.R.; Thom, M.L.; Harlow, E.R.; Gabbett, T.J.; Geletka, B.J.; Hsu, J.J.; Drummond, C.K.; Phelan, D.M.; Voos, J.E. Wearable Technology and Analytics as a Complementary Toolkit to Optimize Workload and to Reduce Injury Burden. *Front. Sport. Act. Living* 2021, 2, 228. [CrossRef]
- Li, S.; Zhang, B.; Fei, P.; Shakeel, P.M.; Samuel, R.D.J. Computational Efficient Wearable Sensor Network Health Monitoring System for Sports Athletics Using IoT. Aggress. Violent Behav. 2020, 101541. [CrossRef]
- Huifeng, W.; Kadry, S.N.; Raj, E.D. Continuous Health Monitoring of Sportsperson Using IoT Devices Based Wearable Technology. Comput. Commun. 2020, 160, 588–595. [CrossRef]
- 5. Altman, A.D.; Helpman, L.; McGee, J.; Samouëlian, V.; Auclair, M.-H.; Brar, H.; Nelson, G.S. Enhanced Recovery after Surgery: Implementing a New Standard of Surgical Care. *CMAJ* **2019**, *191*, E469–E475. [CrossRef]
- 6. Morrison, B.; Kelliher, L.; Jones, C. The Economic Benefits of Enhanced Recovery after Surgery Programmes. *Dig. Med. Res.* 2019, 2, 20. [CrossRef]
- Zhang, C.; Lai, C.-F.; Lai, Y.-H.; Wu, Z.-W.; Chao, H.-C. An Inferential Real-Time Falling Posture Reconstruction for Internet of Healthcare Things. J. Netw. Comput. Appl. 2017, 89, 86–95. [CrossRef]
- Jiang, Y. Combination of Wearable Sensors and Internet of Things and Its Application in Sports Rehabilitation. *Comput. Commun.* 2020, 150, 167–176. [CrossRef]

- 9. Zhao, Y.; You, Y. Design and Data Analysis of Wearable Sports Posture Measurement System Based on Internet of Things. *Alex. Eng. J.* **2021**, *60*, 691–701. [CrossRef]
- Seshadri, D.R.; Li, R.T.; Voos, J.E.; Rowbottom, J.R.; Alfes, C.M.; Zorman, C.A.; Drummond, C.K. Wearable Sensors for Monitoring the Internal and External Workload of the Athlete. *npj Digit. Med.* 2019, 2, 44–59. [CrossRef] [PubMed]
- 11. Bahadori, S.; Immins, T.; Wainwright, T.W. A Review of Wearable Motion Tracking Systems Used in Rehabilitation Following Hip and Knee Replacement. J. Rehabil. Assist. Technol. Eng. 2018, 5, 2055668318771816. [CrossRef]
- 12. Khan, Y.; Ostfeld, A.E.; Lochner, C.M.; Pierre, A.; Arias, A.C. Monitoring of Vital Signs with Flexible and Wearable Medical Devices. *Adv. Mater.* **2016**, *28*, 4373–4395. [CrossRef] [PubMed]
- 13. Rogers, J.; Bao, Z.; Lee, T.-W. Wearable Bioelectronics: Opportunities for Chemistry. Acc. Chem. Res. 2019, 52, 521–522. [CrossRef]
- Calabrese, B.; Velázquez, R.; Del-Valle-Soto, C.; de Fazio, R.; Giannoccaro, N.I.; Visconti, P. Solar-Powered Deep Learning-Based Recognition System of Daily Used Objects and Human Faces for Assistance of the Visually Impaired. *Energies* 2020, 13, 6104. [CrossRef]
- 15. de Fazio, R.; Cafagna, D.; Marcuccio, G.; Minerba, A.; Visconti, P. A Multi-Source Harvesting System Applied to Sensor-Based Smart Garments for Monitoring Workers' Bio-Physical Parameters in Harsh Environments. *Energies* **2020**, *13*, 2161. [CrossRef]
- Pallarés, J.G.; Hernández-Belmonte, A.; Martínez-Cava, A.; Vetrovsky, T.; Steffl, M.; Courel-Ibáñez, J. Effects of Range of Motion on Resistance Training Adaptations: A Systematic Review and Meta-Analysis. *Scand. J. Med. Sci. Sport.* 2021, *31*, 1866–1881. [CrossRef]
- 17. Kumar, R.R.P.; Muknahallipatna, S.; McInroy, J.; McKenna, M.; Franc, L. Real-Time Range of Motion Measurement of Physical Therapeutic Exercises. *J. Comput. Commun.* **2017**, *5*, 19–42. [CrossRef]
- Zhu, Y.; Sun, F.; Jia, C.; Zhao, T.; Mao, Y. A Stretchable and Self-Healing Hybrid Nano-Generator for Human Motion Monitoring. Nanomaterials 2021, 12, 104. [CrossRef] [PubMed]
- Pereira, A.; Guimarães, V.; Sousa, I. Joint Angles Tracking for Rehabilitation at Home Using Inertial Sensors: A Feasibility Study. In Proceedings of the 11th EAI International Conference on Pervasive Computing Technologies for Healthcare, Barcelona, Spain, 23–26 May 2017; Association for Computing Machinery: New York, NY, USA, 2017; pp. 146–154.
- Bell, K.M.; Onyeukwu, C.; McClincy, M.P.; Allen, M.; Bechard, L.; Mukherjee, A.; Hartman, R.A.; Smith, C.; Lynch, A.D.; Irrgang, J.J. Verification of a Portable Motion Tracking System for Remote Management of Physical Rehabilitation of the Knee. *Sensors* 2019, 19, 1021. [CrossRef]
- Wang, A.; Hu, M.; Zhou, L.; Qiang, X. Self-Powered Wearable Pressure Sensors with Enhanced Piezoelectric Properties of Aligned P(VDF-TrFE)/MWCNT Composites for Monitoring Human Physiological and Muscle Motion Signs. *Nanomaterials* 2018, *8*, 1021. [CrossRef]
- Guo, W.; Tan, C.; Shi, K.; Li, J.; Wang, X.-X.; Sun, B.; Huang, X.; Long, Y.-Z.; Jiang, P. Wireless Piezoelectric Devices Based on Electrospun PVDF/BaTiO₃ NW Nanocomposite Fibers for Human Motion Monitoring. *Nanoscale* 2018, 10, 17751–17760. [CrossRef]
- Gaetani, F.; Primiceri, P.; Antonio Zappatore, G.; Visconti, P. Hardware Design and Software Development of a Motion Control and Driving System for Transradial Prosthesis Based on a Wireless Myoelectric Armband. *IET Sci. Meas. Technol.* 2019, 13, 354–362. [CrossRef]
- 24. Kim, H.; Lee, K.; Jo, G.; Kim, J.-S.; Lim, M.T.; Cha, Y. Tendon-Inspired Piezoelectric Sensor for Biometric Application. *IEEE/ASME Trans. Mechatron.* 2021, *26*, 2538–2547. [CrossRef]
- 25. Tognetti, A.; Lorussi, F.; Mura, G.D.; Carbonaro, N.; Pacelli, M.; Paradiso, R.; Rossi, D.D. New Generation of Wearable Goniometers for Motion Capture Systems. *J. NeuroEng. Rehabil.* **2014**, *11*, 56. [CrossRef] [PubMed]
- Li, C.; Liu, D.; Xu, C.; Wang, Z.; Shu, S.; Sun, Z.; Tang, W.; Wang, Z.L. Sensing of Joint and Spinal Bending or Stretching via a Retractable and Wearable Badge Reel. *Nat. Commun* 2021, 12, 2950. [CrossRef]
- Kim, K.; Jung, M.; Jeon, S.; Bae, J. Robust and Scalable Three-Dimensional Spacer Textile Pressure Sensor for Human Motion Detection. *Smart Mater. Struct.* 2019, 28, 065019. [CrossRef]
- Lei, Z.; Wang, Q.; Sun, S.; Zhu, W.; Wu, P. A Bioinspired Mineral Hydrogel as a Self-Healable, Mechanically Adaptable Ionic Skin for Highly Sensitive Pressure Sensing. *Adv. Mater.* 2017, *29*, 1700321. [CrossRef]
- 29. He, X.; Liu, Z.; Shen, G.; He, X.; Liang, J.; Zhong, Y.; Liang, T.; He, J.; Xin, Y.; Zhang, C.; et al. Microstructured Capacitive Sensor with Broad Detection Range and Long-Term Stability for Human Activity Detection. *Npj Flex. Electron.* **2021**, *5*, 17. [CrossRef]
- Dan, L.; Shi, S.; Chung, H.-J.; Elias, A. Porous Polydimethylsiloxane–Silver Nanowire Devices for Wearable Pressure Sensors. ACS Appl. Nano Mater. 2019, 2, 4869–4878. [CrossRef]
- Massaroni, C.; Di Tocco, J.; Lo Presti, D.; Longo, U.G.; Miccinilli, S.; Sterzi, S.; Formica, D.; Saccomandi, P.; Schena, E. Smart Textile Based on Piezoresistive Sensing Elements for Respiratory Monitoring. *IEEE Sens. J.* 2019, 19, 7718–7725. [CrossRef]
- Zhu, L.; Zhou, X.; Liu, Y.; Fu, Q. Highly Sensitive, Ultrastretchable Strain Sensors Prepared by Pumping Hybrid Fillers of Carbon Nanotubes/Cellulose Nanocrystal into Electrospun Polyurethane Membranes. ACS Appl. Mater. Interfaces 2019, 11, 12968–12977. [CrossRef] [PubMed]
- Yun, J.; Cho, H.; Park, J.; Kim, D. Self-Powered and Flexible Triboelectric Sensors with Oblique Morphology towards Smart Swallowing Rehabilitation Monitoring System. *Materials* 2022, 15, 2240. [CrossRef] [PubMed]

- Natta, L.; Guido, F.; Algieri, L.; Mastronardi, V.M.; Rizzi, F.; Scarpa, E.; Qualtieri, A.; Todaro, M.T.; Sallustio, V.; De Vittorio, M. Conformable AlN Piezoelectric Sensors as a Non-Invasive Approach for Swallowing Disorder Assessment. ACS Sens. 2021, 6, 1761–1769. [CrossRef]
- Wilhoite, S.; Mutchler, J.A.; Barry, A.M.; Li, L. Ankle-Knee Initial Contact Angle and Latency to Maximum Angle Are Affected by Prolonged Run. Int. J. Exerc. Sci. 2021, 14, 33–44. [PubMed]
- 36. Kubo, K.; Miyazaki, D.; Yamada, K.; Shimoju, S.; Tsunoda, N. Are the Knee and Ankle Angles at Contact Related to the Tendon Properties of Lower Limbs in Long Distance Runners? *Springerplus* **2016**, *5*, 151. [CrossRef]
- Markström, J.L.; Grip, H.; Schelin, L.; Häger, C.K. Dynamic Knee Control and Movement Strategies in Athletes and Non-athletes in Side Hops: Implications for Knee Injury. *Scand. J. Med. Sci Sport.* 2019, 29, 1181–1189. [CrossRef]
- 38. Buehler, C.; Koller, W.; De Comtes, F.; Kainz, H. Quantifying Muscle Forces and Joint Loading During Hip Exercises Performed With and Without an Elastic Resistance Band. *Front. Sport. Act. Living* **2021**, *3*, 223. [CrossRef]
- Zhao, C.; Jia, C.; Zhu, Y.; Zhao, T. An Effective Self-Powered Piezoelectric Sensor for Monitoring Basketball Skills. Sensors 2021, 21, 5144. [CrossRef]
- Ma, Y.; Ouyang, J.; Raza, T.; Li, P.; Jian, A.; Li, Z.; Liu, H.; Chen, M.; Zhang, X.; Qu, L.; et al. Flexible All-Textile Dual Tactile-Tension Sensors for Monitoring Athletic Motion during Taekwondo. *Nano Energy* 2021, 85, 105941. [CrossRef]
- 41. Li, W.; Lu, L.; Kottapalli, A.G.P.; Pei, Y. Bioinspired Sweat-Resistant Wearable Triboelectric Nanogenerator for Movement Monitoring during Exercise. *Nano Energy* **2022**, *95*, 107018. [CrossRef]
- 42. Liu, W.; Long, Z.; Yang, G.; Xing, L. A Self-Powered Wearable Motion Sensor for Monitoring Volleyball Skill and Building Big Sports Data. *Biosensors* **2022**, *12*, 60. [CrossRef]
- Mariello, M.; Fachechi, L.; Guido, F.; De Vittorio, M. Conformal, Ultra-Thin Skin-Contact-Actuated Hybrid Piezo/Triboelectric Wearable Sensor Based on AlN and Parylene-Encapsulated Elastomeric Blend. *Adv. Funct. Mater.* 2021, *31*, 2101047. [CrossRef]
- Lu, Y.; Liu, Z.; Yan, H.; Peng, Q.; Wang, R.; Barkey, M.E.; Jeon, J.-W.; Wujcik, E.K. Ultrastretchable Conductive Polymer Complex as a Strain Sensor with a Repeatable Autonomous Self-Healing Ability. *ACS Appl. Mater. Interfaces* 2019, *11*, 20453–20464. [CrossRef]
 Donahue, S.R.; Hahn, M.E. Validation of Running Gait Event Detection Algorithms in a Semi-Uncontrolled Environment. *Sensors*
- 2022, 22, 3452. [CrossRef]
- Masihi, S.; Panahi, M.; Maddipatla, D.; Hanson, A.J.; Bose, A.K.; Hajian, S.; Palaniappan, V.; Narakathu, B.B.; Bazuin, B.J.; Atashbar, M.Z. Highly Sensitive Porous PDMS-Based Capacitive Pressure Sensors Fabricated on Fabric Platform for Wearable Applications. ACS Sens. 2021, 6, 938–949. [CrossRef]
- 47. Saponara, S. Wearable Biometric Performance Measurement System for Combat Sports. *IEEE Trans. Instrum. Meas.* 2017, 66, 2545–2555. [CrossRef]
- García-Pinillos, F.; Ramírez-Campillo, R.; Boullosa, D.; Jiménez-Reyes, P.; Latorre-Román, P.Á. Vertical Jumping as a Monitoring Tool in Endurance Runners: A Brief Review. J. Hum. Kinet 2021, 80, 297–308. [CrossRef]
- Worsey, M.T.; Espinosa, H.G.; Shepherd, J.B.; Thiel, D.V. Inertial Sensors for Performance Analysis in Combat Sports: A Systematic Review. Sports 2019, 7, 28. [CrossRef] [PubMed]
- Yang, Y.; Hou, X.; Geng, W.; Mu, J.; Zhang, L.; Wang, X.; He, J.; Xiong, J.; Chou, X. Human Movement Monitoring and Behavior Recognition for Intelligent Sports Using Customizable and Flexible Triboelectric Nanogenerator. *Sci. China Technol. Sci.* 2022, 65, 826–836. [CrossRef]
- Cuperman, R.; Jansen, K.M.B.; Ciszewski, M.G. An End-to-End Deep Learning Pipeline for Football Activity Recognition Based on Wearable Acceleration Sensors. Sensors 2022, 22, 1347. [CrossRef]
- 52. Cosoli, G.; Antognoli, L.; Veroli, V.; Scalise, L. Accuracy and Precision of Wearable Devices for Real-Time Monitoring of Swimming Athletes. *Sensors* 2022, 22, 4726. [CrossRef]
- 53. Chen, L.; Lu, M.; Yang, H.; Salas Avila, J.R.; Shi, B.; Ren, L.; Wei, G.; Liu, X.; Yin, W. Textile-Based Capacitive Sensor for Physical Rehabilitation via Surface Topological Modification. *ACS Nano* 2020, *14*, 8191–8201. [CrossRef]
- 54. Ianculescu, M.; Andrei, B.; Alexandru, A. A Smart Assistance Solution for Remotely Monitoring the Orthopaedic Rehabilitation Process Using Wearable Technology: Re.Flex System. *Stud. Inform. Control* **2019**, *28*, 317–326. [CrossRef]
- Yao, S.; Zhu, Y. Wearable Multifunctional Sensors Using Printed Stretchable Conductors Made of Silver Nanowires. *Nanoscale* 2014, 6, 2345–2352. [CrossRef]
- Ge, J.; Sun, L.; Zhang, F.-R.; Zhang, Y.; Shi, L.-A.; Zhao, H.-Y.; Zhu, H.-W.; Jiang, H.-L.; Yu, S.-H. A Stretchable Electronic Fabric Artificial Skin with Pressure-, Lateral Strain-, and Flexion-Sensitive Properties. *Adv. Mater.* 2016, 28, 722–728. [CrossRef] [PubMed]
- 57. Sheng, L.; Teo, S.; Liu, J. Liquid-Metal-Painted Stretchable Capacitor Sensors for Wearable Healthcare Electronics. *J. Med. Biol.* Eng. 2016, 36, 265–272. [CrossRef]
- Yi, F.; Wang, X.; Niu, S.; Li, S.; Yin, Y.; Dai, K.; Zhang, G.; Lin, L.; Wen, Z.; Guo, H.; et al. A Highly Shape-Adaptive, Stretchable Design Based on Conductive Liquid for Energy Harvesting and Self-Powered Biomechanical Monitoring. *Sci Adv.* 2016, 2, e1501624. [CrossRef] [PubMed]
- 59. Maceira-Elvira, P.; Popa, T.; Schmid, A.-C.; Hummel, F.C. Wearable Technology in Stroke Rehabilitation: Towards Improved Diagnosis and Treatment of Upper-Limb Motor Impairment. *J. NeuroEng. Rehabil.* **2019**, *16*, 20. [CrossRef]
- 60. Li, G.; Dai, K.; Ren, M.; Wang, Y.; Zheng, G.; Liu, C.; Shen, C. Aligned Flexible Conductive Fibrous Networks for Highly Sensitive, Ultrastretchable and Wearable Strain Sensors. *J. Mater. Chem. C* 2018, *6*, 6575–6583. [CrossRef]

- 61. Park, S.W.; Das, P.S.; Chhetry, A.; Park, J.Y. A Flexible Capacitive Pressure Sensor for Wearable Respiration Monitoring System. *IEEE Sens. J.* 2017, 17, 6558–6564. [CrossRef]
- Tan, Y.; Ivanov, K.; Mei, Z.; Li, H.; Li, H.; Lubich, L.; Wang, C.; Wang, L. A Soft Wearable and Fully-Textile Piezoresistive Sensor for Plantar Pressure Capturing. *Micromachines* 2021, 12, 110. [CrossRef] [PubMed]
- 63. de Fazio, R.; Perrone, E.; Velázquez, R.; De Vittorio, M.; Visconti, P. Development of a Self-Powered Piezo-Resistive Smart Insole Equipped with Low-Power BLE Connectivity for Remote Gait Monitoring. *Sensors* **2021**, *21*, 4539. [CrossRef]
- Li, G.-Y.; Li, J.; Li, Z.-J.; Zhang, Y.-P.; Zhang, X.; Wang, Z.-J.; Han, W.-P.; Sun, B.; Long, Y.-Z.; Zhang, H.-D. Hierarchical PVDF-HFP/ZnO Composite Nanofiber–Based Highly Sensitive Piezoelectric Sensor for Wireless Workout Monitoring. *Adv. Compos. Hybrid. Mater.* 2021, *5*, 766–775. [CrossRef]
- Tan, S.; Li, Y.; Fang, F.; Li, L.; Yan, Z.; Liu, D.; Sun, Q. A Microfiber Probe-Based Wearable Sensor for Human Healthcare Monitoring. In Proceedings of the Optical Fiber Sensors Conference 2020 Special Edition (2020), Paper W4.3, Washington, WA, USA, 8–12 June 2020; Optica Publishing Group: Washington, DC, USA, 2020; pp. 1–12.
- Barrias, A.; Rodriguez, G.; Casas, J.R.; Villalba, S. Application of Distributed Optical Fiber Sensors for the Health Monitoring of Two Real Structures in Barcelona. *Struct. Infrastruct. Eng.* 2018, 14, 967–985. [CrossRef]
- Quandt, B.M.; Scherer, L.J.; Boesel, L.F.; Wolf, M.; Bona, G.-L.; Rossi, R.M. Body-Monitoring and Health Supervision by Means of Optical Fiber-Based Sensing Systems in Medical Textiles. *Adv. Healthc. Mater.* 2015, *4*, 330–355. [CrossRef] [PubMed]
- 68. de Fátima Domingues, M.; Tavares, C.; Leite, T.; Alberto, N.; Leitão, C.; Marques, C.; Radwan, A.; Rocon, E.; André, P.A. Fiber Bragg Gratings as E-Health Enablers: An. Overview for Gait Analysis Applications; IntechOpen: London, UK, 2018; ISBN 978-1-78985-352-0.
- 69. Bilesan, A.; Owlia, M.; Behzadipour, S.; Ogawa, S.; Tsujita, T.; Komizunai, S.; Konno, A. Marker-Based Motion Tracking Using Microsoft Kinect. *IFAC-Pap.* **2018**, *51*, 399–404. [CrossRef]
- Bilesan, A.; Behzadipour, S.; Tsujita, T.; Komizunai, S.; Konno, A. Markerless Human Motion Tracking Using Microsoft Kinect SDK and Inverse Kinematics. In Proceedings of the 2019 12th Asian Control Conference (ASCC), Kitakyushu, Japan, 9–12 June 2019; pp. 504–509.
- Guess, T.M.; Razu, S.; Jahandar, A.; Skubic, M.; Huo, Z. Comparison of 3D Joint Angles Measured With the Kinect 2.0 Skeletal Tracker Versus a Marker-Based Motion Capture System. J. Appl. Biomech. 2017, 33, 176–181. [CrossRef]
- 72. Hughes, J.; Iida, F. Multi-Functional Soft Strain Sensors for Wearable Physiological Monitoring. Sensors 2018, 18, 3822. [CrossRef]
- 73. Tolvanen, J.; Hannu, J.; Jantunen, H. Stretchable and Washable Strain Sensor Based on Cracking Structure for Human Motion Monitoring. *Sci. Rep.* **2018**, *8*, 13241. [CrossRef]
- 74. Agarwala, S.; Goh, G.L.; Dinh Le, T.-S.; An, J.; Peh, Z.K.; Yeong, W.Y.; Kim, Y.-J. Wearable Bandage-Based Strain Sensor for Home Healthcare: Combining 3D Aerosol Jet Printing and Laser Sintering. *ACS Sens.* **2019**, *4*, 218–226. [CrossRef] [PubMed]
- 75. Bötzel, K.; Marti, F.M.; Rodríguez, M.Á.C.; Plate, A.; Vicente, A.O. Gait Recording with Inertial Sensors—How to Determine Initial and Terminal Contact. *J. Biomech.* **2016**, *49*, 332–337. [CrossRef] [PubMed]
- Routhier, F.; Duclos, N.C.; Lacroix, É.; Lettre, J.; Turcotte, E.; Hamel, N.; Michaud, F.; Duclos, C.; Archambault, P.S.; Bouyer, L.J. Clinicians' Perspectives on Inertial Measurement Units in Clinical Practice. *PLoS ONE* 2020, *15*, e0241922. [CrossRef] [PubMed]
- 77. Hoareau, D.; Jodin, G.; Chantal, P.-A.; Bretin, S.; Prioux, J.; Razan, F. Synthetized Inertial Measurement Units (IMUs) to Evaluate the Placement of Wearable Sensors on Human Body for Motion Recognition. *J. Eng.* **2022**, 2022, 536–543. [CrossRef]
- Alanen, A.; Räisänen, A.; Benson, L.; Pasanen, K. The Use of Inertial Measurement Units for Analyzing Change of Direction Movement in Sports: A Scoping Review. Int. J. Sport. Sci. Coach. 2021, 16, 1332–1353. [CrossRef]
- 79. Homayounfar, S.Z.; Andrew, T.L. Wearable Sensors for Monitoring Human Motion: A Review on Mechanisms, Materials, and Challenges. *SLAS Technol.* **2020**, *25*, 9–24. [CrossRef]
- Peng, Y.; Yang, N.; Xu, Q.; Dai, Y.; Wang, Z. Recent Advances in Flexible Tactile Sensors for Intelligent Systems. Sensors 2021, 21, 5392. [CrossRef]
- Yao, S.; Swetha, P.; Zhu, Y. Nanomaterial-Enabled Wearable Sensors for Healthcare. Adv. Healthc. Mater. 2018, 7, 1700889. [CrossRef]
- 82. Zhao, T.; Li, T.; Chen, L.; Yuan, L.; Li, X.; Zhang, J. Highly Sensitive Flexible Piezoresistive Pressure Sensor Developed Using Biomimetically Textured Porous Materials. *ACS Appl. Mater. Interfaces* **2019**, *11*, 29466–29473. [CrossRef]
- Yao, S.; Yang, J.; Poblete, F.R.; Hu, X.; Zhu, Y. Multifunctional Electronic Textiles Using Silver Nanowire Composites. ACS Appl. Mater. Interfaces 2019, 11, 31028–31037. [CrossRef] [PubMed]
- 84. Sun, R.; Carreira, S.C.; Chen, Y.; Xiang, C.; Xu, L.; Zhang, B.; Chen, M.; Farrow, I.; Scarpa, F.; Rossiter, J. Stretchable Piezoelectric Sensing Systems for Self-Powered and Wireless Health Monitoring. *Adv. Mater. Technol.* **2019**, *4*, 1900100. [CrossRef]
- Allataifeh, A.; Al Ahmad, M. Simultaneous Piezoelectric Noninvasive Detection of Multiple Vital Signs. Sci. Rep. 2020, 10, 416. [CrossRef] [PubMed]
- Chorsi, M.T.; Curry, E.J.; Chorsi, H.T.; Das, R.; Baroody, J.; Purohit, P.K.; Ilies, H.; Nguyen, T.D. Piezoelectric Biomaterials for Sensors and Actuators. *Adv. Mater.* 2019, *31*, 1802084. [CrossRef] [PubMed]
- Stadlober, B.; Zirkl, M.; Irimia-Vladu, M. Route towards Sustainable Smart Sensors: Ferroelectric Polyvinylidene Fluoride-Based Materials and Their Integration in Flexible Electronics. *Chem. Soc. Rev.* 2019, 48, 1787–1825. [CrossRef] [PubMed]
- Maity, K.; Garain, S.; Henkel, K.; Schmeißer, D.; Mandal, D. Self-Powered Human-Health Monitoring through Aligned PVDF Nanofibers Interfaced Skin-Interactive Piezoelectric Sensor. ACS Appl. Polym. Mater. 2020, 2, 862–878. [CrossRef]
- 89. Wang, Z.L.; Wang, A.C. On the Origin of Contact-Electrification. Mater. Today 2019, 30, 34–51. [CrossRef]

- Zhang, N.; Tao, C.; Fan, X.; Chen, J. Progress in Triboelectric Nanogenerators as Self-Powered Smart Sensors. J. Mater. Res. 2017, 32, 1628–1646. [CrossRef]
- 91. Yi, F.; Zhang, Z.; Kang, Z.; Liao, Q.; Zhang, Y. Recent Advances in Triboelectric Nanogenerator-Based Health Monitoring. *Adv. Funct. Mater.* **2019**, *29*, 1808849. [CrossRef]
- 92. Kim, D.W.; Lee, J.H.; Kim, J.K.; Jeong, U. Material Aspects of Triboelectric Energy Generation and Sensors. *NPG Asia Mater.* 2020, 12, 6. [CrossRef]
- 93. Choi, W.; Yun, I.; Jeung, J.; Park, Y.S.; Cho, S.; Kim, D.W.; Kang, I.S.; Chung, Y.; Jeong, U. Stretchable Triboelectric Multimodal Tactile Interface Simultaneously Recognizing Various Dynamic Body Motions. *Nano Energy* **2019**, *56*, 347–356. [CrossRef]
- De Fazio, R.; De Vittorio, M.; Visconti, P. A BLE-Connected Piezoresistive and Inertial Chest Band for Remote Monitoring of the Respiratory Activity by an Android Application: Hardware Design and Software Optimization. *Future Internet* 2022, 14, 183. [CrossRef]
- 95. Castañeda, J.J.; Ruiz-Olaya, A.F.; Lara-Herrera, C.N.; Roldán, F.Z. Knee Joint Angle Monitoring System Based on Inertial Measurement Units for Human Gait Analysis. In Proceedings of the VII Latin American Congress on Biomedical Engineering CLAIB 2016, Bucaramanga, Santander, Colombia, 26–28 October 2016; Torres, I., Bustamante, J., Sierra, D.A., Eds.; Springer: Singapore, 2017; pp. 690–693.
- Bavan, L.; Surmacz, K.; Beard, D.; Mellon, S.; Rees, J. Adherence Monitoring of Rehabilitation Exercise with Inertial Sensors: A Clinical Validation Study. *Gait Posture* 2019, 70, 211–217. [CrossRef] [PubMed]
- 97. Filippeschi, A.; Schmitz, N.; Miezal, M.; Bleser, G.; Ruffaldi, E.; Stricker, D. Survey of Motion Tracking Methods Based on Inertial Sensors: A Focus on Upper Limb Human Motion. *Sensors* **2017**, *17*, 1257. [CrossRef] [PubMed]
- 98. De Fazio, R.; Stabile, M.; De Vittorio, M.; Velázquez, R.; Visconti, P. An Overview of Wearable Piezoresistive and Inertial Sensors for Respiration Rate Monitoring. *Electronics* **2021**, *10*, 2178. [CrossRef]
- Hughes, C.M.L.; Louie, A.; Sun, S.; Gordon-Murer, C.; Belay, G.J.; Baye, M.; Zhang, X. Development of a Post-Stroke Upper Limb Rehabilitation Wearable Sensor for Use in Sub-Saharan Africa: A Pilot Validation Study. *Front. Bioeng. Biotechnol.* 2019, 7, 322. [CrossRef]
- 100. De Fazio, R.; De Vittorio, M.; Visconti, P. Innovative IoT Solutions and Wearable Sensing Systems for Monitoring Human Biophysical Parameters: A Review. *Electronics* **2021**, *10*, 1660. [CrossRef]
- McGrath, M.J.; Scanaill, C.N. Sensing and Sensor Fundamentals. In Sensor Technologies: Healthcare, Wellness, and Environmental Applications; McGrath, M.J., Scanaill, C.N., Eds.; Apress: Berkeley, CA, USA, 2013; pp. 15–50. ISBN 978-1-4302-6014-1.
- 102. Wang, T.; Green, R.; Guldiken, R.; Wang, J.; Mohapatra, S.; Mohapatra, S.S. Finite Element Analysis for Surface Acoustic Wave Device Characteristic Properties and Sensitivity. *Sensors* **2019**, *19*, 1749. [CrossRef]
- Koncar, V. Structural Health Monitoring of Processes Related to Composite Manufacturing. In *Smart Textiles for In Situ Monitoring of Composites;* Koncar, V., Ed.; The Textile Institute Book Series; Woodhead Publishing: Sawston, UK, 2019; pp. 295–381. ISBN 978-0-08-102308-2.
- 104. Keaton, J.R. Young's Modulus. In *Encyclopedia of Engineering Geology*; Bobrowsky, P.T., Marker, B., Eds.; Encyclopedia of Earth Sciences Series; Springer International Publishing: Cham, Switzerland, 2018; pp. 955–956. ISBN 978-3-319-73568-9.
- 105. Soloman, S. Sensors Handbook, 2nd ed.; McGraw Hill: Boston, MA, USA, 2009; ISBN 978-1-59693-307-1.
- 106. Barhoum, A.; Altintas, Z. Advanced Sensor Technology: Biomedical, Environmental, and Construction Applications; Elsevier Science: Amsterdam, The Netherlands, 2022; ISBN 978-0-323-90222-9.
- 107. O'Reilly, M.; Duffin, J.; Ward, T.; Caulfield, B. Mobile App to Streamline the Development of Wearable Sensor-Based Exercise Biofeedback Systems: System Development and Evaluation. *JMIR Rehabil. Assist. Technol.* 2017, 4, e9. [CrossRef]
- 108. Salehi, S.; Bleser, G.; Reiss, A.; Stricker, D. Body-IMU Autocalibration for Inertial Hip and Knee Joint Tracking. *EAI Endorsed Trans. Pervasive Health Technol.* **2015**, *2*, e3.
- Yao, S.; Vargas, L.; Hu, X.; Zhu, Y. A Novel Finger Kinematic Tracking Method Based on Skin-Like Wearable Strain Sensors. *IEEE Sens. J.* 2018, 18, 3010–3015. [CrossRef]
- 110. Yang, G.; Deng, J.; Pang, G.; Zhang, H.; Li, J.; Deng, B.; Pang, Z.; Xu, J.; Jiang, M.; Liljeberg, P.; et al. An IoT-Enabled Stroke Rehabilitation System Based on Smart Wearable Armband and Machine Learning. *IEEE J. Transl. Eng. Health Med.* 2018, 6, 1–10. [CrossRef]
- Wu, J.; Huang, J.; Wang, Y.; Xing, K.; Xu, Q. Fuzzy PID Control of a Wearable Rehabilitation Robotic Hand Driven by Pneumatic Muscles. In Proceedings of the 2009 International Symposium on Micro-NanoMechatronics and Human Science, Nagoya, Japan, 9–11 November 2009; pp. 408–413.
- Lipovský, R.; Ferreira, H.A. Hand Therapist: A Rehabilitation Approach Based on Wearable Technology and Video Gaming. In Proceedings of the 2015 IEEE 4th Portuguese Meeting on Bioengineering (ENBENG), Porto, Portugal, 26–28 February 2015; pp. 1–2.
- 113. Chen, T.; Lum, P.S. Hand Rehabilitation after Stroke Using a Wearable, High DOF, Spring Powered Exoskeleton. In Proceedings of the 2016 38th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (EMBC), Orlando, FL, USA, 16–20 August 2016; pp. 578–581.
- 114. Li, G.; Cheng, L.; Gao, Z.; Xia, X.; Jiang, J. Development of an Untethered Adaptive Thumb Exoskeleton for Delicate Rehabilitation Assistance. *IEEE Trans. Robot.* 2022, *38*, 3514–3529. [CrossRef]

- 115. Guo, L.; Zhang, B.; Wang, J.; Wu, Q.; Li, X.; Zhou, L.; Xiong, D. Wearable Intelligent Machine Learning Rehabilitation Assessment for Stroke Patients Compared with Clinician Assessment. J. Clin. Med. 2022, 11, 7467. [CrossRef]
- 116. Ren, Y.; Wu, Y.-N.; Yang, C.-Y.; Xu, T.; Harvey, R.L.; Zhang, L.-Q. Developing a Wearable Ankle Rehabilitation Robotic Device for In-Bed Acute Stroke Rehabilitation. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2017, 25, 589–596. [CrossRef] [PubMed]
- Zhang, P.; Zhang, Z.; Cai, J. A Foot Pressure Sensor Based on Triboelectric Nanogenerator for Human Motion Monitoring. Microsyst. Technol. 2021, 27, 3507–3512. [CrossRef]
- Mao, Y.; Yue, W.; Zhao, T.; Shen, M.; Liu, B.; Chen, S. A Self-Powered Biosensor for Monitoring Maximal Lactate Steady State in Sport Training. *Biosensors* 2020, 10, 75. [CrossRef] [PubMed]
- He, J.; Xiao, P.; Lu, W.; Shi, J.; Zhang, L.; Liang, Y.; Pan, C.; Kuo, S.-W.; Chen, T. A Universal High Accuracy Wearable Pulse Monitoring System via High Sensitivity and Large Linearity Graphene Pressure Sensor. *Nano Energy* 2019, 59, 422–433. [CrossRef]
- Tao, L.-Q.; Zhang, K.-N.; Tian, H.; Liu, Y.; Wang, D.-Y.; Chen, Y.-Q.; Yang, Y.; Ren, T.-L. Graphene-Paper Pressure Sensor for Detecting Human Motions. ACS Nano 2017, 11, 8790–8795. [CrossRef]
- 121. Li, Y.; Miao, X.; Niu, L.; Jiang, G.; Ma, P. Human Motion Recognition of Knitted Flexible Sensor in Walking Cycle. *Sensors* 2020, 20, 35. [CrossRef] [PubMed]
- Mao, Y.; Zhang, W.; Wang, Y.; Guan, R.; Liu, B.; Wang, X.; Sun, Z.; Xing, L.; Chen, S.; Xue, X. Self-Powered Wearable Athletics Monitoring Nanodevice Based on ZnO Nanowire Piezoelectric-Biosensing Unit Arrays. *Sci. Adv. Mater.* 2019, *11*, 351–359. [CrossRef]
- Coyle, S.; Morris, D.; Lau, K.-T.; Diamond, D.; Moyna, N. Textile-Based Wearable Sensors for Assisting Sports Performance. In Proceedings of the 2009 Sixth International Workshop on Wearable and Implantable Body Sensor Networks, Berkeley, CA, USA, 3–5 June 2009; pp. 307–311.
- 124. Tesconi, M.; Tognetti, A.; Scilingo, E.P.; Zupone, G.; Carbonaro, N.; De Rossi, D.; Castellini, E.; Marella, M. Wearable Sensorized System for Analyzing the Lower Limb Movement during Rowing Activity. In Proceedings of the 2007 IEEE International Symposium on Industrial Electronics, Vigo, Spain, 4–7 June 2007; pp. 2793–2796.
- 125. University of Idaho University of Idaho—Physical Rehabilitation Movements Data Set (UI-PRMD). Available online: https://webpages.uidaho.edu/ui-prmd/ (accessed on 1 February 2023).
- 126. Capecci, M.; Ceravolo, M.G.; Ferracuti, F.; Iarlori, S.; Monteriu, A.; Romeo, L.; Verdini, F. The KIMORE Dataset: KInematic Assessment of MOvement and Clinical Scores for Remote Monitoring of Physical REhabilitation. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2019, 27, 1436–1448. [CrossRef]
- 127. Kolykhalova, K.; Camurri, A.; Völpe, G.; Sanguineti, M.; Puppo, E.; Niewiadomski, R. A Multimodal Dataset for the Analysis of Movement Qualities in Karate Martial Art. In Proceedings of the 2015 7th International Conference on Intelligent Technologies for Interactive Entertainment (INTETAIN), Turin, Italy, 10–12 June 2015; pp. 74–78.
- 128. UCI Machine Learning Repository: Daily and Sports Activities Data Set. Available online: https://archive.ics.uci.edu/ml/ datasets/daily+and+sports+activities (accessed on 1 February 2023).
- Reyes-Ortiz, J.L.; Anguita, D.; Ghio, A.; Oneto, L.; Parra, X. Human Activity Recognition Using Smartphones Data Set. Available online: https://archive.ics.uci.edu/ml/datasets/Human+Activity+Recognition+Using+Smartphones (accessed on 1 February 2023).
- 130. Ghorbani, S.; Mahdaviani, K.; Thaler, A.; Kording, K.; Cook, D.J.; Blohm, G.; Troje, N.F. MoVi: A Large Multipurpose Motion and Video Dataset. *PLoS ONE* **2021**, *16*, e0253157. [CrossRef]
- Hausdorff, J.M.; Purdon, P.L.; Peng, C.-K.; Ladin, Z.; Wei, J.Y.; Goldberger, A.L. Gait in Aging and Disease Database. Available online: https://archive.physionet.org/physiobank/database/gaitdb/ (accessed on 1 February 2023).
- 132. Isaac, E.R.H.P.; Elias, S.; Rajagopalan, S.; Easwarakumar, K.S. Trait of Gait: A Survey on Gait Biometrics. *arXiv* 2019, arXiv:1903.10744.
- 133. Prakash, C.; Kumar, R.; Mittal, N. Recent Developments in Human Gait Research: Parameters, Approaches, Applications, Machine Learning Techniques, Datasets and Challenges. *Artif. Intell. Rev.* **2018**, *49*, 1–40. [CrossRef]
- do Nascimento, L.M.S.; Bonfati, L.V.; Freitas, M.L.B.; Mendes Junior, J.J.A.; Siqueira, H.V.; Stevan, S.L. Sensors and Systems for Physical Rehabilitation and Health Monitoring—A Review. Sensors 2020, 20, 4063. [CrossRef]
- Patel, S.; Park, H.; Bonato, P.; Chan, L.; Rodgers, M. A Review of Wearable Sensors and Systems with Application in Rehabilitation. J. Neuroeng. Rehabil. 2012, 9, 21. [CrossRef]
- 136. Li, R.T.; Kling, S.R.; Salata, M.J.; Cupp, S.A.; Sheehan, J.; Voos, J.E. Wearable Performance Devices in Sports Medicine. *Sport. Health* **2016**, *8*, 74–78. [CrossRef]
- 137. Yang, S.; Li, C.; Wen, N.; Xu, S.; Huang, H.; Cong, T.; Zhao, Y.; Fan, Z.; Liu, K.; Pan, L. All-Fabric-Based Multifunctional Textile Sensor for Detection and Discrimination of Humidity, Temperature, and Strain Stimuli. *J. Mater. Chem. C* 2021, 9, 13789–13798. [CrossRef]
- Wei, Y.; Li, X.; Wang, Y.; Hirtz, T.; Guo, Z.; Qiao, Y.; Cui, T.; Tian, H.; Yang, Y.; Ren, T.-L. Graphene-Based Multifunctional Textile for Sensing and Actuating. ACS Nano 2021, 15, 17738–17747. [CrossRef] [PubMed]
- Perovich, L.; Mothersill, P.; Farah, J.B. Awakened Apparel: Embedded Soft Actuators for Expressive Fashion and Functional Garments. In Proceedings of the 8th International Conference on Tangible, Embedded and Embodied Interaction, Munich, Germany, 16–19 February 2014; Association for Computing Machinery: New York, NY, USA, 2014; pp. 77–80.

- 140. Tong, Y.; Kucukdeger, E.; Halper, J.; Cesewski, E.; Karakozoff, E.; Haring, A.P.; McIlvain, D.; Singh, M.; Khandelwal, N.; Meholic, A.; et al. Low-Cost Sensor-Integrated 3D-Printed Personalized Prosthetic Hands for Children with Amniotic Band Syndrome: A Case Study in Sensing Pressure Distribution on an Anatomical Human-Machine Interface (AHMI) Using 3D-Printed Conformal Electrode Arrays. *PLoS ONE* 2019, 14, e0214120. [CrossRef]
- 141. Jeong, S.Y.; Shim, H.R.; Na, Y.; Kang, K.S.; Jeon, Y.; Choi, S.; Jeong, E.G.; Park, Y.C.; Cho, H.-E.; Lee, J.; et al. Foldable and Washable Textile-Based OLEDs with a Multi-Functional near-Room-Temperature Encapsulation Layer for Smart e-Textiles. *Npj Flex. Electron.* 2021, *5*, 15. [CrossRef]

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.